# Social Capital and Burmese Migrant Domestic Workers in Thailand

Author: Ganon Koompraphant Supervisor: Monica Lindberg Falk



**Master's Programme in Asian Studies** 

Spring semester 2014

## **Table of Content**

Chapter 1: Introduction	1
- Aims	2
- Research Question	2
- Limitations and Demarcation	2
- Disposition	3

Chapter 2: Research Methodology and theoretical frameworks
- Research Method4
Stance and Strategy4
Method for Collecting Information4
<ul> <li>Interview</li></ul>
Sample Selection and limitation6
Data Analysis and Presentation7
Ethical Consideration7
- Theoretical Framework
Theorizing Social Capital in Transnational Context
Social Capital8
Social Capital and Transnationalism
Theorizing migrant's perspective13
• Subjective Experience
• Agency

Chapter 3: Background and Contextual Information	15
Internationalization of care economy and feminization of migration	15
• The context of labor migration from Burma in Thailand	17
Victim or agent?	20
• The cases of migrants from Burma in Thailand	21
Labour movement and social capital in the context of transnational workers	22
• Social Capital and Social Network of Migrant Workers in Thailand	24
Chapter 4: Migrants' experience and the meaning of social capital	26
The Informants	26
Social Capital and Migration Process	28
Social Capital and Working Life in Thailand	31
Fellow Villager	31
• Employer	35
Religious Institution	37
Development Organization	37
Trust and cost of social capital in the context	
of being domestic migrants workers	38
Gender, Ethnicity and Class	40
• Gender	40
• Ethnicity	42
• Class	43
Solidarity and Social Capital as Public Good	45
Social capital and migrant's agency	46

Chapter 5: Concluding Remark	50
Reference	54
Appendix	.61

## Abstact

The purpose of this thesis is to explore and understand the significant of social networks for Burmese domestic migrant workers who live in Thailand. In order to achieve that, the thesis employs the concept of social capital to scrutinize different social networks in the migrants' lives. The analysis in the thesis has been done both in larger level focusing on the role and important of social capital to the migration process and lives; as well as, in micro level, examining different social capital from the perspectives of the migrants.

The data has been collected in Thailand by using semi-structure interviews and observations. The thesis concludes that social capital is crucial to both the process of migration and lives as migrant workers in Thailand. On the micro level, the specific socio-cultural context of migrants' and the migrants' own agency are also shaping the way the migrants build and use their social capital. The thesis has also found that, despite its importance, social capital has negative sides. In addition, the migrants' choices and access to various social capitals in Thailand are limited.

## Acknowledgement

Writing this thesis have made me realized the important of social capital in a much more profound way than when I read the concept as a theory. Without the kindness of the following people, I will not be able to complete this thesis.

I would like to thank Monica Lindberg Falk for her advice, encouragement and kindness both in the academic and personal matters. I'm truly grateful and honored to have a chance to work under your supervision.

In addition, I would like to express my gratitude to Gripen scholarship and AnandaMahidol foundation for an opportunity to study in Sweden.

The thesis could not be written in the first place if my informants have not trusted me and told me about their life. Although you might not be able to read this, thank you so much for sharing part of yourself with me, I truly hope that you all will reach the goals and dreams you have.

I could not express enough thanks for my contacts who helped me during my field work. Due to ethical consideration I could not mention your name here, but you know who you are. I would like to express a deep gratitude toward all my friends especially Amp, Ariel, Fah, Jivaza and Sofia who endure me and give me encouragements when I'm in the worst version of myself. And a special thank to Maria for her proof-reading.

Lastly, I would like to thank my family, who not because of norm or obligation but pure love, give me indefinite supports.

## **Chapter 1: Introduction**

Due to the change in the structure of the Thai labor market and the regional disparity, migrants from Burma<sup>1</sup> have become the largest group of foreign workers in Thailand (Huguet and Chamratrithirong, 2011: 1 - 2).

Since I was a teenager at the end of the 1990s in Bangkok, I have experienced that the domestic workers in my neighborhood have been dominated by women from Burma. One by one, my relatives started to employ migrant domestic workers from Burma. Apart from their unique Thai accents, how my relatives praise them as much better workers than Thais, and how I was told to tell no stranger that we employed Burmese migrants, for me they are no different than Thai maids.

It was not until I had a chance to be a migrant myself, as an exchange student in the United States, that I realized how difficult their migration must have been. I also realized that I knew nothing about the domestic workers from Burma at all. My feelings of anxiety, fear, alienation and helplessness as a migrant exchange student initiated many questions about the transnational domestic workers in Thailand. My migration process has been largely influenced and assisted by a social network whose responsibilities ranged from recommending studies agency to being a network of support that help me get through all the difficulties. Therefore, it seems that social capital mobilized from the network is crucial even in my middle class regular<sup>2</sup> migration experience (Putnam, 1995: 67).

Comparing my migration experience to those migrant workers from Burma that I have encountered, they seem to live in a context that needs even more assistance than me. Many of the domestic migrants that I have encountered came to work in Thailand irregularly, without proper language skills, were not aware of the formal channels of help and were not accustomed to communication technology. Thereof, it seems that the migrants' social networks and personal relationship could play important roles in their migrant worker lives. Thus, I started to wonder how do their social networks contribute to and shape their migration

<sup>&</sup>lt;sup>1</sup> Regarding the usage of Burma/Myanmar, I have decided that I will use the term Burma to refers to the country as this is the name all of my informants use. In order to avoid confusion, I will use the word "ethic Burmese" to describes people that identify themselves with this ethnic group. The word "Burmese" will be used to refer to nationality.

<sup>&</sup>lt;sup>2</sup> The thesis will use the term '**regular**' instead of '**legal**' and '**irregular**' instead of '**illegal**' as I believes in the same stance as United Nation agencies that the term illegal "carries a criminal connotation" (International Organization for Migration - IOM,2014)

experiences. This thesis, employing the concept of social capital, attempts to address this problem.

#### Aims of the thesis

This thesis aims to investigate the meaning of social capital in the life of female migrants from Burma who work as domestic workers in Thailand from their own subjective experience and perspective. This study also provides account of the migrants on their experience living in Thailand. In addition, to a lesser extent, the migrants' relationship with their social network will be analyzed on an individual level. The thesis is a qualitative case studies using semi – structured qualitative interviews as the main method of information collection; some aspects of life story interview technique will be used as well.

#### Research question:

The main research question: what is the significant of social capital in the life of Burmese female domestic migrant workers in Thailand?

#### Sub –Questions

How do social networks facilitate the migration process?

How do social networks influence the life as female domestic migrant workers?

How do gender, ethnicity and class shape and influence social capital usage?

How is the domestic workers' agency articulated?

#### Limitations and Demarcation

The thesis uses a qualitative approach with limited and specific samples. All of the informants in this thesis live in the Bangkok Metropolitan area, are female and are able to speak Thai. The sample cannot be claimed to represent all the experience of migrant domestic workers in Thailand. Also, the argument and information in this thesis is based on informants' subjective narrative and experience. The presentation of information is also concentrated only on information that is related to the concept of social network/ social

capital. This means several interesting points from the migrants' account of their life that are less related to the thesis' main theoretical framework are not included even though they are interesting and display complex dimensions of the migrants' life.

The thesis mainly focuses on exploring the role of migrants' social networks using the concept of social capital. Therefore, other aspects of Thai – Burma female labor migration such as the influence of structural levels will not be addressed. Also, within the concept of social capital itself, this thesis doesn't emphasis much on trust building process within the networks or how social capital encourages democratic norms as emphasized in Putnam (1995, 2002). All the analytical topics presented in the research are mostly based on the theme identified through interviews, some points might be emphasized more or emphasized less than other migration literatures.

## **Disposition**

The thesis comprises of five chapters. Chapter 1 introduces the aim, research questions and the importance of the research topics. Chapter 2 discusses the selection of research methods and the theoretical framework for the thesis. Chapter 3 reviews the current literatures and debates on female migration, as well as providing the contextual information on female migration from Burma to Thailand. Chapter 4 presents the experience of the informants as transnational domestic migrants workers in Thailand and analyzes the experience according to the chosen theories. Chapter 5 offers a concluding remark.

## **Chapter 2 : Methodology and Theoretical Framework**

#### Research Method

## Research Stance and strategy

This study employs qualitative methods with a social constructivist approach which holds that there are multi – dimensional aspect of reality and that reality is socially constructed (Moses and Knutsen, 2007: 194). It means that the thesis also acknowledges the inter-subjectivities of involved agents on the research production especially the role of myself as the thesis writer and the migrants from Burma as the subjects of the studies both in the data gathering and data creation process, as well as in the interpretation of data (Mahoney,2007 :574). Therefore, addressing reflexivity and being aware of bias and my involvement in the research process is important and a reflexive writing technique will be employed in this thesis (Bryman, 2004: 500).

This research employs a strategy of case studies. As put by Yin (2003:1,7), this strategy is suitable for investigating the complex phenomenon that research cannot manipulate the research setting. Thus, this is a proper strategy for the research topic. Case studies can be criticized for "providing little basis for scientific generalization" (Yin, 2003:10). Nevertheless, it is important to highlight that the aims of the research is not to find samples that represent scientific generalization of population / variables, but to analyze the interviews as samples for theoretical case studies of the lives of migrant domestic workers in Thailand (Moses and Knutsen:2007, 140).

## Method for Collecting Information

• Interview

The qualitative semi-structural interview with topic specific questions focusing on the role of social networks and their capital according to each migrant's experience will be the main research method (Bryman, 2008: 438 -439). The advantage of using semi-structured interviews is that, while they allow me to obtain the information according to the focus of the

thesis, they also give some space for the migrants' own voices and points of view (Bryman, 2008 :464).

Some perspectives from life story research, which emphasizes on how events shape individual's lives by letting subject speaks of their experience, are used in this thesis (Bryman, 2004: 321). Due to time limits and the research topic, however, this thesis will not conduct a proper life story "narrative interview" (Jovchelovitch and Bauer, 2000: 59), which enables interviewees to speak flexibly about their past on the aspects that they consider to be important without specific focus. The thesis instead will employ the life story research framework to shape the way interviews are conducted, as well as to shape interview questions. Life story research can provide an insight to how individuals understand and interpret the world (Bryman, 2008:440). Applying life story methods will allow me to explore migrants' perspectives and opinions on their social network and how they perceive the network's social capital in accordance to their own experience. Moreover, it will also mean that the data received from interviews will be treated as subjective truth and understanding of informants rather than as a source of pure objective truth (Behar ,1990: 226 - 227). This element seems to be appropriate to the aims of the research which attempts to understand the meaning of social capital according to the migrants' subjective experience.

#### • Participants Observation

Although interviews are the main data-collecting method in this research, I have also taken the role of "Non-Participating Observer with Interaction" (Bryman, 2012:444). This means that I have observed and interacted with the informants but not truly participated in their activities. Most of the interactions are through interviews and small talk not through participation in their daily works. Through my experience living in a household where domestic workers are employed and visiting a home with domestic workers from Burma several times, I am familiar with the context and attitude of Thai employers, their relationships with the domestic workers and the work/routines of domestic migrants themselves. I have also observed how the informants interact with me, someone who represents a different class, gender and ethnicity from them as I am a Thai male who comes from the same social circle as their employers. Apart from this, I have also attended two religious activities that were recommended by the informants and also observed the interaction of migrants who attend vocational Sunday school (Section on sample selection below provides additional details on the research site). I have observed the process of activities that migrants participate in and the interaction during the activities. The data from participant observation in this research is used to provide contextual information on the life of domestic migrant workers in Thailand.

#### Sample Selection and limitation

I have interviewed eleven domestic workers from Burma. Although I did not intend to be specific regarding the gender of the interviewees, all of domestic workers I have interviewed are female. The age of the interviewees varies from nineteen to approximately forty years old. All of them currently work as domestic workers and most often this has been their only employment. The length of the interviews was approximately one to two hours. All of the informants currently live in the Bangkok Metropolitan area. Most of the interviews have been recorded.

I have also attended three other activities (once at each activity). The first was Yiguandao Movement (In Thai: ลัทธิอนุตครธรรม), which one of my interviewees was attending and had great faith on. Another one was the activity at a Buddhist temple where ethnic Karen migrants from a certain region in Burma come together to perform a religious ceremony together. The ceremony also involves a fund raising project for sending Karen monks to be educated abroad. At the ceremony, I had a chance to talk to the domestic workers who participated in the activities there, as well as, a chance to interview the monk who helps coordinate these activities. The final activity was the Sunday vocational school for migrants set up by the Thai Action Committee for Democracy in Burma (TACDB) which teaches language and computer skills to migrants from Burma in Thailand.

Snowball sampling method (Bryman, 2012: 202) has been used to identify both the informants and the religion activities. I met most informants through my own contacts in Thailand. The criteria for selection are that the interviewee must be able to communicate in Thai, partly due to my limited language skills. Nevertheless, the ability to speak Thai also indicates that the migrant has already slightly adapted to Thai society and is able to have some interaction with Thais. Thus, it could be assumed that Thai context is more embedded to their life than in the lives of migrants who can't speak Thai and mainly interact with their own ethnic group. These migrants are therefore expected to have more types of social capital.

Another criteria is they must not currently work in the border town such as Mae Sot as the context of migrants' live there should be clearly different than the rest of the countries.

Selecting samples through snowball method, though, can be criticized as non scientific (Bryman, 2008: 184 - 185). However, as pointed out in the discussion on case studies, the research does not claim that the interviewees could represent the whole population of domestic workers from Burma in Thailand, but instead aims to use the information as an example for theoretical analysis.

#### Data Analysis and Presentation

After the interviews were conducted, they were transcribed and summarized. After that the contents were analyzed through "thematic analysis" methods where the information were grouped, analyzed and presented according to the patterns found in the data (Vaismoradi, 2013: 400).

The analysis will also employ the multi- voices style as indicated in Mahoney (2007: 574) in which all voices involved in the researching topic, including the voice of myself as the researcher and those of the interviewees, are acknowledged. In the thesis, I will attempt to be transparent in my subjectivity role in producing the content. Nevertheless, it must be acknowledged that my voice will still be the authoritative voice in the research since I am the one who chooses what text and which part of the interviews will be presented and the data is also put into the sequence I have designed and analyzed by the theoretical concept I have chosen (Mahoney, 2007: 583).

#### Ethnical Consideration

The research follows the ethical consideration section in "Guidelines for the

Master's Thesis", published by the Centre for East and South-East Asian Studies, Lund University (2013). In order to avoid potential "harm to participants', the following measures were taken. All of the informants in this thesis are "informed consent" (Bryman, 2004: 511). This means that the purpose of the studies and the interviews along with, the potential readers and potential areas where the study may be disseminated, have been clearly explained before any interview attempt. The names of the informants are kept anonymous to keep their privacy and safety (Bryman, 2004: 513). To prevent "invasion of privacy", the interviewees are informed that they can choose to terminate the interview or their involvement in this project whenever they wish (Bryman, 2012:142).

Although ethical guidelines advise that my research process should not affect the "subjects' relationship with their environment" (Bryman, 2012: 136), I have faced the dilemma of how much I should inform my informants or shape their understanding especially when I have found that their labor rights are violated. Regarding this point, I have decided to provide them with information in accordance to their rights but left the informants to decide on their own how to respond to the information.

## Concept and Theoretical Framework

The concept of Social capital by Putnam (1993, 1995, 2002) and Faist (2000) is used to examine the significant of social network in the life of domestic migrants' workers in Thailand. This thesis also employs the concept of transnationalism to examine social capital in the context of transnational migration.

While social capital could explain the facilitators that enable migrants to move to and live in a different context, it doesn't explain how their network and the social capital they have shape their migration experience from their perspective. Therefore, the concept of subjective experience (Brittle, 2000: 111) and domestic workers as an active agent (Nurchayati, 2011:480; Ueno, 2013: 37 – 38; Johnson, 2011: 462) will be used in order to comprehend their experience and their relationship with social networks/capital in a more subtle way.

## Theorizing social capital in transnational context

• Social Capital

According to Putnam (1995: 67) the definition of social capital is "features of social organization such as networks, norms, and social trust that facilitate coordination and cooperation for mutual benefit". Basically, social capital is a network, norm or association both that could be used to provide supports to individual such as family, peer group or ethnic based organization (Woolcock and Narayan, 2000: 3; Putnam, 2002:10). The belief that social capital could be used to enhance the individual comes from the notion that every social network has value and this value can be used for benefit (Putnam, 2002: 6). For example, farmers in the same village could initiate labor exchange in cultivating and harvesting which could save time and cost for both parties (Putnam, 1993: 163).

The benefits of social capital have not been clearly defined yet. In the text by Putnam (2002:6), he mentions that there is some attempt to calculate the benefit of social capital into "cash value" but at the same time there are also many literatures that focus on the effect of social capital on things that aren't numerical measurable such as mental condition or even democracy.

Thus, it is useful to employ the concept of social capital proposed by migration sociologist, Thomas Faist (2000), in order to frame the benefit of social capital. First, it should be informed that Faist identifies features of social capital in addition to Putnam. He reveals that social capital is created based on ties (Faist: 2000, 101 -102). The ties could be either a "social tie" or a symbolic tie. A social tie involves people with whom one associates and has an interpersonal relation. It can have various sizes, forms and closeness and it could be a relationship with family, an association, or an authority. On another hand, symbolic ties are the ties that are fostered by shared identity which could connect people together even without actual interaction such as belonging to the same ethnic community or religion.

Based on this notion, Faist (2000: 102) combines the concept of social capital between those of Bourdieu which concentrates on social capital as a capital for individuals and Putnam who emphasizes on the network and norm dimensions of social capital. He proposes that there are two features of social capital's benefit. The first feature is social capital as a resource that an individual or group can use to achieved their specific goal. This means that the benefit of social capital varies depending on how it aims to be used (Faist, 2000: 102). The second feature is social capital as a mechanism for collaboration as it can link different social actors together; thus being beneficial by providing a platform for collaboration as well as advancing the group's benefit (Faist, 2000: 103). The benefit of social capital can be both at the individual and public level (Putnam, 2002:7). The benefit in each level is mutually enhancing. While an individual can mobilize resources from the network, being a part of and having a social tie to the network and its norms also strengthens the common norm/ practices that can be translated into public benefit and vice versa (Putnam, 2002: 7). For example, neighborhood watch programs that attempt to reduce crime rates in communities can positively affect individuals who live in that community by increasing the general safety of the community. Moreover, this also means that if social capital provides public benefits, the benefits can be transferred to someone who has not directly "invested" in and assisted the group or network but still belongs to the network in some ways through a "social tie" or

"symbolic tie" such as those in the neighbourhood who have not participated in crime reduction cooperation but still receive the benefits from the practice (Putnam, 2002: 7; Faist, 2000, 101 -102).

The public benefit can play a tremendous role in shaping the whole society. For example, if social trust, social engagement, norms and collective action between different networks are developed on a wider level, they could provide a platform for democratic institutions to emerge (Putnam, 1993: 182).

However, there is a unique feature of social capital's benefit that needs to be pointed out. As Faist (2000: 103, 118) mentioned, even though social capital can be invested and its benefits can be expected, social capital and its benefits are more a "by-product of transaction" with the function of facilitating rather than the actual result of investment. This resonates with the way Putnam frames the element of social capital. According to his article, social capital can be used, accumulated and mobilized through everyday interactions such as sharing information, debating, helping each other, talking or even casting votes (Putnam, 1995: 67 - 68).

Putnam (1993:103) illustrates that networks and social capital are strengthened and beneficial if only members of networks keep their obligations and follow the expected behaviors. To achieve that, it means the network should be held by and embedded with strong norms that make the individual acknowledge the reciprocal practice within their social capital circle (Putnam,1993:168). Thereof, trust within networks is increased. Once there is more social trust, individuals are more willing to contribute to others as they feel sure that they will receive the same treatment in return (Putnam, 1993: 168). Therefore, collective action and platforms for collective action are enhanced. Apart from norms and obligations mentions by Putnam (1993), Faist also adds that solidarity within ties, such as a feeling of nationalism, are also a mechanism that supports the cooperation of social capital (Faist, 2000: 109).

With the "norms of generalized reciprocity", it means that having social capital not only brings benefits, sometimes individuals must also offer something in order to keep social capital. For example, one is expected to help their peer when their request for help; Woolcock and Narayan (2000: 3) conceptualize this process as "cost" of social capital. It should also be noted that social capital isn't always used for positive benefit, especially when it is framed as resources that could support a different goal (Faist, 2000: 103). Social capital, thus, could have an exclusionary nature and be used to enhance conflicts and discrimination such as in organizations like Klu Klux Klan (Putnam, 2002:9).

Putnam (2002: 11 -13) points out that social capital could be divided into several different typologies depending on which features of social capital are viewed as follows:

1. The first type is "formal" and "informal" (Putnam, 2002: 10), such as association (formal) versus family (informal).

2. "Thick" and "thin" / "strong tie" and "weak tie" (Putnam, 2002: 10) Thick/ Strong ties basically means a network that one have more associates and closeness with such as co-workers and the one that there is less chance to associate with such as people that one walk pass in the apartment.

3. "Inward" looking and outward "looking" (Putnam, 2002: 11). Inward looking means the social capital whose only concern is for people within the network while outward looking is the capital that attempt to provide public good. It should be noted that this doesn't means outward looking social capital will actually provide more public benefit as the characteristic and goal of each social capital should also be evaluated additionally.

4. "Bridging" and "bonding" (Putnam, 2002: 11). Bonding social capital means social capital that is created by interaction between people who are alike and share some similar aspects such as race, class or gender. While bridging social capital is created by an interaction of people who are considered to be different. Putnam seems to stress the importance of differences between these two capitals as it is expected that having bridging capital will provide more "positive externality effect" than bonding capital (Putnam, 2002: 11). Too much concentration on bonding capital, for example, could lead to intergroup conflict such as ethnic conflict (Putnam, 2002: 12). Nevertheless, same social capital often consist of both bridging and bonding elements such as the union where people share economic status but not ethnicity or religion (Putnam, 2002: 12).

In addition to all the foregoing, it should be reminded that while social capital theorists often focus on the positive effects or negative external effects of social capital and network, I agree

with points mentioned in the thesis of Panthaloet (2012: 16) that there is not so much investigation of the negative internal effects of social capital and network and how it might, as well, constrain ones agency.

• Social Capital and Transnationalism

Transnationalism is a perspective that challenges the previous notion that migrants are "uprooted" (Brettell, 2000: 104), according to the perspective, migrants actually maintain their tie with both their native and their receiving countries both in terms of movement and in terms of economic, social and cultural context, migrant also could be the linkage of change both for the sending and receiving society.

As observed by Faist (2000: 122), migrants often still have both "social ties" and "symbolic ties" to their home community. The form of transnational ties are various, they could be on a micro level such as family to business partners or Diasporas communities (Faist, 2000: 202 - 208). The ties bridge two spaces in different borders and provide a platform for the flow of "multiple transactions of ideas, monetary resources, goods, symbols, and cultural practices" that transcend the national border (Faist, 2000:196). This notion is similar to what Levitt (1998) conceptualized as "social remittance" where she finds out that "norms, practices, identities and social capital (and cultural)" is exchanged between host and origin communities (Levitt and Lamba-Nieves, 2002: 3). As for social capital, for example, Levitt (1998: 935 -936) points out social capital found abroad could be used to enhance status at home and migrants also often have fulfilled the expectation of their ties in order to keep the transnational social capital.

In contrast to the concept of "uprooted" where the ties to origin communities are abridged, migrants according to transnationalism can migrate, emigrate, and re-migrate between national boundaries when the condition is permitted and mobilization is suited to their goal (Faist, 2000: 200).

Many transnational network theorists, have used the notion of transnational social networks/ social capital in order to fulfill the gap that previous theories on migration that either focus on cost-benefit analysis of individual or larger social economic context could not answer; for example, why two communities situated in similar contexts have a totally different rate of migration (Faist, 2000: 94; Panthaloet, 2012: 14; Brettell: 2000, 104).

According to Faist (2000), to answer the question above, social capital also serves an important function that shapes the migration process.

The role of social capital can influence and assist individuals' decisions on whether to migrate (Faist, 2000: 122-123; 144 -145; 201), provide a platform for chain migration or even the create a culture of migration for people who have ties with the pioneer migrant, consolidate transnational linkages that surpass first generations of migrant and change the chain migration into transnational social space where linkages are kept in the succeed generation, and lastly, help migrants adapt and cope with the new context in their destination.

Last but not least, it should be noted that the concept of "social remittance" (Levitt, 1998) as well as how potential migrants are influenced by social ties who are migrants themselves shows that the influence between migrants and their social ties are mutual (Faist, 2000: 122). They both are influenced by their own social capital and they also influence their social capital. However, in my observation, it seems that the transnational perspective often focuses on how migrants' change and the transformation of context influences their origin communities rather than how their communities continues to shape them after their mobility.

#### Theorizing migrant's perspective

• Subjective Experience

The recent trend of research on women experience has emphasized on the subjective and individuality view point of their experience (Brittle, 2000: 111). The perspective is developed to challenge the theoretical uniformity of how one experiences oppressions; for example, arguing that women's oppression according to main stream feminism cannot represent the experience of all women (Brittle, 2000: 111; Herr, 2014: 2).

The principle of this concept is that subjectivity is created through the intersection of multidimension social categories such as gender, class and ethnicity (Nash, 2008:2). Through examining the intersection of these labels, the complex and context-specific experience of each subject can be seen. This notion, according to Herr, means the viewpoint of every female should be respected even if it is not adhered to in mainstream theory such as feminism's beliefs (2014: 6). Females should also be viewed as active agents who have a capacity to control their life rather than being total victims (Charrad, 2010: 519).

In her studies on female Dominican migrants in Boston, Pessar (1995) provides a good example of why understanding the experience of female migrant in a specific context is important. She finds out that some Dominican female migrants decided that cooperation with unions to improve their working condition was not worthwhile as it could cost them a job and limit their space to home where their life fill with boredom because they would have nothing to do and have no other relationship/interaction except with their family (Pessar, 1995: 45).

Nevertheless, this perspective is not left without critics. There are numerous debates regarding whether focusing on subjective experience in relation to contexts means that the political struggles to change structural inequality are ignored (Coleman, 2009: 7).

#### • Agency

According to Charrad (2010: 519) agency refers to the "capacity to initiate change or to commit oneself to a transformative or challenging course of action". She also points out that agency should be examined in relation with the contextual information surrounding each individual (Charrad,2 010: 519). Under the concept of agency, although the structure might provide a suppressive context that constrains individuals, there is always room for individuals to act and find ways to resist, - even in the most oppressive context (Johnson, 2011: 460). While the concept of agency often has a connotation with an action to challenge existing hierarchies and enhance independence and self determination, sometimes it is also possible that one would makes a choice that creates even more limited space for one own agency (Charrad, 2010: 519). Thus, agency should be examined in relation to specific hierarchies in different contexts.

This study frames agency in the context of migrant domestic workers who have to deal with employers, family and the hierarchical structure of the state where migrants, with limited rights compared to citizens and the nature of their work, are often put to the bottom of this hierarchy. The studies of Nurchayati (2011: 480) and Johnson (2011) show that in order to achieve their goal, migrant domestic workers try to adjust the situation, manage trouble, find methods to deal with abuses and constantly attempt to bargain with anyone that has authority over them (such as their employers) in intelligent and complex ways.

## **Chapter 3: Background and Contextual Information / Literature Review**

#### Internationalization of care economy and feminization of migration

The social- economic gap and hierarchy between countries and regions has been a condition for the globalization of the care economy. The globalization of care economy or "global care chain" means the increasing demand for migrant workers form relatively poorer countries in the global south to work in relatively richer countries on the work sector that Lutz's book "The new maid: transnational and care economy" (2011: 4, 26) called "the three Cs: cooking, caring and cleaning" or what Parrenas's article (2001: 364) called "emotional labor". These sectors of work that traditionally are often considered to be female's unpaid duties to provide in the private sphere. It is assumed that this growing demand reflects the change of social-economic conditions and gender relations in the richer destination country itself, as population in those countries, especially female, has been deeper integrated into the labor market outside the home and the need for foreign workers to replace their duties in the house has increased (Lutz, 2011: 24; Resurreccion, 2010: 38). Brooke (2009: 4) pointed out female jobs outside home require full commitment from labor force, but does not provide enough support to compensate for the loss of workers who perform domestic works at home.

This practice, thus, partly stimulates the recent trend of 'feminization of migration' where the number of female migrant workers is rapidly increasing to match or even outnumber male migrants (Castles and Miller, 2009: ). For example, as mentioned in Brooke's article (2009: 4), at least 65% of migrant workers from the Philippines, Indonesia and Sri Lanka are female. According to Parrenas's articles (2001: 362 -363) this migration trend seems to be a reversal from the traditional trend where males are the ones who usually migrate to work abroad in order to fulfill their 'bread winner' role. This shift of gender migration also changes the perspective on the role of females in migration from females as 'the followers' of male migranst to females who take initiative of their own life/migration experience instead (Panthaloet, 2012:1; Kusakabe and Pearson, 2010).

In her article on the social impact of female migration, Piper (2008: 12) argues that the focus on gender aspects of migration is important as generally the migration experience of male and female migrant workers could be different. She points out that labor migration in Asia is often shaped by clear gender based division of labor where most of female migrants work in "social reproduction" jobs such as care work or jobs that required delicacy, while most men often have jobs in pure labor sectors such as construction (Piper, 2008: 1292). Therefore, it could be said that some aspects of migrant experience are gender specific.

It is shown that in many cases, the life and living conditions of those who work in female jobs such as migrant domestic workers are prone to risky situations. In their article on transnational domestic workers in Singapore, Rahman, Yeoh and Huang (2005) indicate that working and living in the inferior position under the private space that is controlled by the employer seems to be the nature of domestic workers' job. This means it is likely that they must comply with working conditions and rules that are in accordance to their employers' need. For example, they can be prone to abuses such as long working hours to suit their employers' life style without payment substitution. In some case, the migrants even have to sacrifice their own cultural practices in order to comply with the demand of employers. For example, some Muslim girls are not allowed to practice their daily prayers (Rahman, Yeoh and Huang, 2005). In addition, according to, Brooke (2009:4) and Lutz (2011: 19) as domestic works are conducted in the private sphere, there is not so much chance for migrant to contact any other people beside their employer, thus making it more difficult for domestic migrant workers to ask for help. There are also cases of sexual abuse towards domestic migrant workers by their employers (Nurchayati, 2011; Johnson, 2011).

The article by Rahman, Yeoh and Huang (2005:245) and the article by Chin (2005) on transnational domestic migrants in Malaysia both point out that domestic workers can also be made in public sphere as the others who have tendency to threaten the security and order of the host country; thus they often have the policies where migrant are welcomed to stay only as the 'supply' for employers' demand but are not welcome as part of the community. In Malaysia, for example, the otherness of migrant workers has been solidified by a law preventing transnational domestic workers from being married to Malaysian citizens (Chin, 2005: 272). The notion of dangerous other has lead to more control of employers toward migrants' private affairs. For instance, some employers in Singapore think that allowing migrants to be out of the home and associate with people from the migrant's community without their close supervision would be prone to what they consider as "unethical" activity

such as having sexual activities. For this reason, many migrants do not have a day-off (Rahman, Yeoh and Huang, 2005: 245). Furthermore, labor laws in many countries often exclude domestic workers or other informal female jobs such as sex workers from formal legal protection and recognition, thus, increasing the risk of them being exploited (Brooke, 2009:4; Lutz, 2011:19, Chin, 2005: 272, Briget, 2010: 64 – 65; Ally, 2005).

Not to mentioned that domestic migrant workers often have to leave their family behind only to work for the benefit of others' family in richer countries as the law often doesn't permit the family of the workers to migrate and live with them in the destination country or workers' may not have enough resources to afford moving the entire family (Luzt, 2011: 24). This can sometimes cause migrants emotional distress with the loneliness and guilt of not being able to take care of the family and perform the expected duties of mother/ daughter (Brooke, 2009: 8).

Piper's (2008: 1296) article on the development aspect of feminization migration also reveals that the trend of migrant labor diversification in terms of country of origin can also cause negative effects for female migrant workers' status as the employer may prefer workers from a country where workers are not well-aware of workers' rights and are more obedient toward the employers. For example, there is less demand for maids from Philippine in richer countries in East Asia as the Philippine state is more active in protecting the rights of their migrant workers.

However, it also should be mentioned as in line with the arguments pointed out in the article examining transnational care work in the context of Asia by Huang, Thang and Toyota (2012). The article (Huang, Thang and Toyota, 2012: 130 - 132) argues that the characteristic of transnational care work should not be viewed as universal and based on economy alone as cultural context is equally important in order to understand the process and complexity of care work in different settings. Thus the fact that care works can be conceptualized differently should always be in consideration.

• The context of labor migration from Burma in Thailand

The general information of migrant workers in Thailand seems to be conformed to the migration trends mentioned above. According to Myat's article (2010: 2-3) overviewing Burma – Thai migration, Thailand's low-skilled labor shortage, economic gap, and relatively better living quality compared to its neighbor countries are factors that draw many

transnational migrant workers to work in Thailand. On the other hand, Burma's long border with Thailand, its relatively large population, as well as, political and economic instability are push factors driving people from Burma to seek jobs in Thailand (Myat, 2010: 2-3). While there is no exact number of migrant workers from Burma in Thailand, it could be assumed that they are Thailand's biggest group of foreigners and mostly, at first, come to work in Thailand irregularly (Myat, 2010: 1). This can be seen through the statistic of unskilled migrant workers from Burma who have come to Thailand irregularly and are in the process of regularizing their status<sup>3</sup>. According to IOM (2013), as of 25 July 2013, the number of migrant workers from Burma who entered Thailand irregularly and are still in or have completed the regularization process is around 1.2 million; in comparison, the number of unskilled migrant workers who come to work in Thailand through official state-run programs is only 36,650 (IOM, 2013). Moreover, it needs to be mentioned that there may be a substantial amount of migrants from Burma who are working irregularly in Thailand and are not included into this statistic.

As pointed out in IOM's migration report, Thai state policies and attitude toward transnational migrants has been shifting back and forth between acknowledging the need for a cheap labor force to sustain its labor intensive export oriented economy (Hall, 2011: 24) and the concern for national security (Archavanitkul and Hall, 2011: 67). Thus as pointed out in the article by Toyota (2005: 293) on domestic workers from Burma in Thailand, this often results in reluctant migration policies that accept low skilled migrants to work in Thailand but enforce strict laws to control any 'potential problems' which often jeopardize migrant workers' basic rights and provide an opportunity for authorities to extort from the labor smuggling industry between Thailand and Burma.

It is not surprising that the system of migrant management and registration in Thailand is complicated and the details of procedures are constantly changing. The main idea of Thai migration policy is to regularize the irregular migrants who are already working in Thailand while trying to develop channels for importing migrants from Burma to work in Thailand regularly. At the same time there are no policies or plans for this group of migrants to be long-term residents in Thailand as migrant only should be in Thailand because of the demand from Thai employers. The last round of regularization was taking place in 2013 which will give workers from Burma a passport that is valid for six years and rights to work in Thailand

<sup>&</sup>lt;sup>3</sup> This is the only statistic dealing with the number of low skilled migrant workers in Thailand

for four years, as well as a work permit. To complete the process workers must also pay for health insurance and medical check-ups as well. The workers who have completed the regularization process will receive the same basic labor rights as Thai laborers (IOM, 2013). According to my interview with a colleague from the development organization (February, 2014) I used to work with, the latest problem in the regularization process is Thailand's political situation. The political unrest and having an acting government disables Thailand's ability to extend the working period of migrant workers who entered the regularization process in 2009 and many of who's permission to work in Thailand is going to be expired in April or May 2014. It is expected that around 700,000 migrants will be affected by this problem<sup>4</sup>.

According to Toyota (2005:289), employing female domestic workers has always been a practice in Thai society; however, domestic work is viewed negatively by Thai society because of its connotation with a low social position and because of the view that it is a job that no people who have a better choice would choose to do. It is not surprising that with the expansion of the Thai economy there is a shortage of domestic workers supply and Thai households have started to hire migrant workers instead (Toyota, 2005: 290).

Even with a constant attempt to regularize domestic migrant workers, to a certain extent it seems that the lives of migrant workers in Thailand who work in the domestic sphere are relatively more vulnerable than other migrants; similar to the overall trend, domestic workers in Thailand receive less protection from labor laws and the nature of work leaves them in a vulnerable position. It should be noted that the law to guarantee the basic labor rights of domestic workers in Thailand was just authorized in 2012; however, even within this law domestic workers is still receive less protection than other types of laborers (Matichon, 2012). The comprehensive survey studies on domestic migrant workers from Burma in Thailand by Panam, Zaw, Caouette, and Punpuing , (2004: xii - xiv) also point out that as most domestic migrant workers in Thailand live-in with their employers without so much chance to interact with their own peers, they live in conditions that could easily lead to being exploited for instance having to work in long hours without proper compensation.

Toyota (2005: 297) argues that the exploitation of transnational domestic workers in Thailand is not only caused by economic motives but also shaped by cultural practices as housemaids who live just for food, clothes and accommodation without pay has always been a tradition of

<sup>&</sup>lt;sup>4</sup> Kanokporn, 22/02/2014

Thailand. And the nature of the relationship between domestic workers and their employers is not often based on economic but reciprocity systems where the domestic workers are not considered as economic laborers who have rights but as someone who found assistance from people with higher status. The result of my interviews which will be presented later seems to confirm that this kind of practice is still embedded in the mind set of some employers.

#### *Victim or agent?*

There have been numerous debates on whether low-skilled female migrants such transnational domestic workers should be viewed as victims or active agents. With the example of the vulnerable conditions mentioned above, it is quite reasonable that domestic migrants are viewed as victims. As pointed out in the articles of Nurchayati (2011:480), Ueno (2013: 37 - 38) and Johnson (2011 :462) much of the literature on migrant domestic workers and reports published by human rights organizations often depict migrant domestic workers as victim who are exploited by multi - actors in multi-level and dimensions including the current world political economic structure. According to Ueno (2013: 37)'s article, even the literature on migrant domestic workers' relationships often depicts the "love drain" aspect of migrant life where a mother or wife has traumatic experience as her care/love is drained away from her own family to the destination countries' family instead.

Thus, it is important to point out that many literatures also explore how working abroad has provided various opportunities and experience that are not available for the migrants at home, which in the end could strengthen their agency. Their new role as 'bread winner', for example, could challenge the patriarchal gender relation at home (Luzt, 2011: 21, 25). Furthermore, working in the domestic sphere doesn't always mean that domestic workers will take a passive role in the relationship with the employers and solely rely on them, as indicated in Lutz (2011: 186). There are also cases where migrant domestic workers and employers mutually rely on each other such as when employers also establish trust with migrants and do want them to stay at the job. The articles of Nurchayati (2011) and Johnson (2011) also depict that migrants do not passively accept their challenging circumstances and try to resist as well as negotiate to change their situation and gain more power. Ueno's article (2013:38) also challenges the "love/care drain" concept by indicating that numerous migrants attempt to have romantic relationships in the destination country (Singapore in her case) due

to the benefit for their migration experience or for their social mobility both in term of legal and social status. Paul's article (2009) also mentions how their experience being domestic migrant workers have provided some migrants knowledge and skills that enable them to change their circumstances to the context where they can be treated better as workers, such as moving from being maids in Saudi Arabia to working in western countries.

Therefore, as Lutz (2011: 185) points out, the boundary between 'active agent' and 'victim' is not clearly distinguished and migrant workers can be viewed in both positions. These two positions, in reality, are inseparable and should be analyzed together in order to fully comprehend the situation of each migrant worker (Luzt, 2011: 185)

• The cases of migrants from Burma in Thailand

Being 'victims' seems to be the representation that dominates the literature on Migrants from Burma in Thailand especially in those which are written by development agencies. According to Wai (2004 in Bhumiprabhas and Kerdmongkol, ed.), Koetsawang (2001) and Human Rights Watch (2010), the abuses that migrant workers face in Burma such as forced labor or violence from war have left them no other choice but to come to work in Thailand. For their migration experience in Thailand, the three literatures (Wai, Bhumiprabhas and Kerdmongkol, 2004 ; Koetsawang , 2001 ; Human Rights Watch, 2010) indicate that they have received various forms of abuses from various stakeholders including their employers, labor agents, brokers, trafficking rings, and Thai authorities, as well as, inadequate access proper criminal justice processes in all migration processes.

In the studies of Panam, Zaw, Caouette, and Punpuing (2004) which focus specifically on migrant domestic workers from Burma, the position of migrants as a 'victims' is also depicted. They (Panam, Zaw, Caouette, and Punpuing ,2004), point out various problems that domestic workers face in their migration experience from the struggling life in Burma to countless exploitation from diverse sectors in Thailand. They also highlight awareness raising, right protection and empowerment as policy recommendations to authorities.

However, although it seems not to be the initial intention of this research (Panam, Zaw, Caouette, and Punpuing ,2004), it is one of a few studies on migrants from Burma published by a Thai organization that not only presents the migrant as victims but also as 'active agents'. For example, there is a discussion that even if some of the female migrants face a lot of obstacles in Thailand, they still prefer to stay in Thailand rather than going back to Burma

as they want to escape pressure from family to conform to the role of a good daughter. There is also a discussion on the intention of these migrant to improve and advance themselves with their migration experience.

Panthaloet's (2012) thesis on Lao female migrants who works in Karaoke bars in Thailand also points out the unclear boundary and complexity between being a victim or being an agent. For example, some Laos women she interviewed have been tricked into working at karaoke bars, but at a certain time, adapted to the working conditions there and became willing to be exploited on some level rather than go home in order to secure the benefit of being workers (Panthaloet, 2012: 172 - 173).

In addition, it should also be pointed out that migrants' agency is not only exercised in the work-related sphere but also in general everyday life, as indicated in Kerdmongkol's thesis (2011). Although migrants from Burma are strictly controlled by Thai authority's migrant workers management system, they are still able to find space to exercise their resistance. For example, through regular activities such as shopping, migrants workers create a normal life space that is not conformed to the controlled space created by Thai state.

#### Labour movement and social capital in the context of transnational workers

What is the relationship between transnational labor migration and social capital, how do migrants accumulate and acquire their social capital, and how does social capital shape the migration experience in various aspects seems to be the main question in the existing research on migration and social capital.

Multiple publications (Hopkins, 2011; Chamlee-Wright and Storr, 2011), have stressed the importance of collective identity, such as sharing place of origin or shared grievance/ traumatic experiences, as the base for migrants to build social capital; s. For instance, as illustrated in Cambell (2012), the multi –ethnic migrants from Burma in Thailand have achieved cohesion between people with different ethnic and religious background because of the shared frustrated migration experience. In addition to that, working in the same place and being in the same situation can also be a platform for mobilization of social capital. One example is given by Anderson (2010: 62 -63) through his depiction of the domestic migrant workers' movement in United Kingdom (UK) which started as an anti-Marcos regime movement by Pilipino migrants, but turned into an organization advocating for domestic

workers regardless of nationality through the contact that Pilipino maids had with migrants of other nationalities who worked with them.

However, it is not necessary for migrants to build social capital or prefer bonding social capita (people who are similar to them) over bridging capital (people who are different). Sometime migrants will choose to enhance their status in one network even if it means that they will need to sacrifice their role in another social network; even if the network is bonding social capital that share a certain degree of similarity with migrants themselves. For example, some migrant domestic workers from the Philippines refusing to help their countrymen to find jobs as it might ruin their reputation with employers (Paul, 2013). Or in some cases, it seems that shared identity is not a vital part in social capital building for migrants. As indicated in the field studies of Kim (2013: 357), one of the most important factors for migrants in determining who to create networks with is the social position of that person rather than shared identity.

Nevertheless, based on their field reports, Nurchayati (2011), Johnson (2011) and Ueno's (2013) s articles still indicate that fellow workers from the same countries of origin (and villages) seem to be the main social networks/capital that domestic migrant workers have abroad. These fellow workers seem to be the sources of information and mental support such as exchanging experience on harassment they have received, and being a social safety net that could give them assistance in emergency situations (Nurchayati ,2011; Johnson, 2011).

Ueno (2013) demonstrated that having lovers or boyfriends, regardless of nationality, can be useful social capital for migrants both physically and psychologically; interestingly, the content of her article also reveals a more complicated account on relationships among migrant workers where there is constant competition to be better and gain higher society status among fellow migrants as well such as having Caucasian boyfriends who assert their identity more than fellow workers.

In terms of formal institutional social capital, the articles of Ally (2005: 187 -188) and Ford (2004: 101 -102) state that, generally, it is quite difficult for domestic migrant workers to have institutionalized collective action, such as having active labor unions, due to their nature of work, more private relationships with employers and how they are viewed as not being "real workers" which create structural and legal barriers for collective action. Ally (2005) also mentions that, even within unions, domestic workers have always been treated with lower status and are given less priority because the union often thinks that domestic workers

cannot be mobilized, thus domestic workers often have to rely on other types of organizations such as NGOs which according to her sometime view workers as their "client victim" (189).

Nevertheless, Ally (2005: 200 -201) asserts that the notion that domestic workers unions cannot be organized is a myth and there are some examples of successful domestic workerbased organizations due to a shift of strategy from servicing labor to advancing political goals and labor rights; as well as, conducting the activities on public space rather than the workplace.

In spite of that, Ford (2004) offers another perspective on domestic workers and unions. Although there is no formal labor union of domestic workers, the boundary between different kinds of organizations working with migrant workers such as unions and NGOs is more flexible than given credit for in the literature. For example, there are numeral domestic worker's networks and organization that are created through the support from NGOs (Ford, 2004,111-112).

Anyhow, as illustrated in Anderson's (2010) article on the migrant domestic workers' movement in the UK, successfully mobilizing may not be enough to make a significant change ; the strategy for using the mobilized power and the level of activeness is quite important as well. As shown by the example of domestic workers' organization movement in the UK such as 'Waling Waling' which had took the strategy to develop migrant to be active citizen who fighting for rights and recognition (Anderson, 2010: 61-63). With this strategy, the organization seems to have had some success as the UK has started a policy to regularize migrant domestic workers. However, Anderson (2010, 69) also reveals that since the shift of the state's policy the organization and the members are less active and are more concerned about their case or about bringing their family members to UK than with conducting a politically motivated movement.

#### • Social Capital and Social Network of Migrant Workers in Thailand

There are few studies on social capital in the context of transnational migrant workers in Thailand and most of the studies seem to emphasize more on the concept of social networks among migrant themselves or ethnic communities rather than on social capital. Nevertheless, some of them have provided background information and interesting observation on low skilled transnational migrant's network in Thailand. In the studies of Kerdmongkol (2011) on Pa - O migrants (from Burma) in Thailand, he concludes that the transnational communities of Pa - O migrants are created in order to give the support needed for survival and adaptation to life in Thailand. The communities are tied together in three characteristics which are kinship and peer based community, ethnic based community and a community that is tied by shared experience of being transnational migrants.

However, the role of social capital in the migration experience does not only limit the social capital in the destination country; on another hand, social capital in country of origin is also essential for migration experience. According to a survey by Pearson and Kusakabe (2012 :167), 50% of their sampled migrants from Burma have their extended family helping them to raise children, as many cannot effort the cost or commitment of raising children by themselves in Thailand. Nonetheless, in this case, migrants also have to contribute to these network, such as sending back remittance, in order to keep ties (Pearson and Kusakabe, 2012 : 167).

Anyhow, as pointed out in the theoretical section, social capital or networks do not only provide benefits but they can also be the source of conflicts and exclusion as well. As Panthaloet's thesis on Laos migrants in Thailand (2012: 173) argues that the nature of migrant's social network is quite dynamic rather than static. For example, as the case in her studies suggests, the characteristic of relationships within the network of Laos' migrants is often shifting between supportive and conflicting. This can sometimes entirely diminish the ties to the social network for some migrants themselves. For example, a Laos migrant who was found by her peer to be a spy for police received tremendous social sanctions that caused her to leave her job at a Karaoke shop (Planthaloet, 2012: 174).

Also, it should be noted that how and what kind of social capital is built does not depend on individual agency alone, but is also shaped by other factors such as gender expectations. Thus, for instance, the reasons for migrants to remit money home might not only be for mutual economic benefit, but also to fulfill the duties as good daughters as well.

## Chapter 4: Migrants' experience and the meaning of social capital

*Me (Ganon): Why don't you move to the US. since you have a right to do as a refugee? Mom (my informant): In Thailand if I have no works I can contact someone, there I have no one.*<sup>5</sup>

This chapter presents the experience of the informants as transnational domestic workers. It also presents an analysis of their experiences according to the selected theoretical concepts. The analysis displays the role of social capital in the context of the lives of female transnational migrant domestic workers. It also attempts to shows the migrants' perspective on their relationship with social networks.

## The Informants (See Appendix 1 for background information on the informants)

- All interviewees from this section are female
- All of them work as domestic workers at present
- All of them work in Bangkok Metropolitan Area
- Except for Aye (not sure less than 5 years) and Nam (around 8 years), all interviewees have been working in Thailand for more than ten years.
- All of them have not completed high school

Wan – 40 years old. She is the only interviewee who knew me personally prior to the interviews. She now lives with her second husband and their daughter in a room provided by her employers. She has another older daughter who lives in Burma with her previous husband. Wan was more open to me than any other informant. She was a migrant worker in Burma before she came to Thailand.

Ngam – 38 years old. Ngam's job is to take care of her employer's factory but she doesn't have to do regular cleaning work. She only has to clean her employer's home twice a week.

<sup>&</sup>lt;sup>5</sup> Mom, 09/03/2014

Her husband is a high ranked worker in the factory and they live separated from her employers' house. Ngam has quite a good reputation in the eyes of a Thai person who recommend her to me; she told me that Ngam has proper manners and is very articulate when talking. She comes from the same village as Wan and was an internal migrant worker in Burma before coming to Thailand. She has two children. She is extremely interested in religion.

Aye – 19 years old. Aye is a young maid who is now working for my relative. She, unlike my other interviewees, was born and grew up in a refugee camp. She doesn't have much working experience. She is single and has no boyfriend. In my opinion she is the most naïve of all of the interviewees.

Nid – 38 years old. Nid is the only ethnic Burmese informant. . She doesn't live in her employer house but lives with her husband in their own rented room. Her husband works as a construction worker at the temple near her work place.

Moei – 24 years old. I met Meoi at the migrant Sunday vocational school which I found through my NGO contact. The director of the school introduced Moei to me. As a 24 year old, she is the second youngest of my interviewees, but her life and work experience in Thailand could be compared to the interviewees in their thirties since she has lived in Thailand for more than ten years and has strong agency. When I interviewed her, she told me her own narrative rather than waiting for me to ask questions. She is very aware about her rights as a worker in Thailand and dreams of social mobility. She is single and doesn't want to be in a romantic relationship.

Cherry – 35 years old. I met her at the vocational school as well and she was also introduced to me by the director of the school. She already has completed courses there but still comes to school to do volunteer work. Among my interviewees she is well aware and knowledgeable about her rights as a migrant worker in Thailand. She treats her job professionally and is much empowered. She is working with Western employers. She is single and has no boyfriend.

Nam – 40 years old. . Her current employer is the only employer she has had in Thailand. Compared to other informants of her age, she is less knowledgeable about Thailand and has less fluent language skills. She is a widow. Bear – 29 years old. Bear works as a baby sitter. I met Bear twice, but the first time was just to give a brief introduction regarding my thesis and the interview. The second meeting was for full interview. Compared to many interviewees, Bear is quite confident as she seems not to be anxious about my presence at all (she always came late to our appointments and told me her needs directly). She has a son whom she brought back to Burma while she and her husband are still working in Thailand.

Apple -29 years old. I had a short interview with her as she was not finished with her daily work and seemed to want to return to work. She is single.

Mom - 30 - 40 years old. She is Aye's mother. She was born in Burma but also grew up in refugee camp. She has two daughters; one of them is still living in the refugee camp. She, her husband and her daughters now live separately and can only meet on Sunday when they have a day off from work. She has been working as a maid since she was teenager. Her husband is working in a furniture factory.

Baitong – 44 years old. She comes from the same village as Wan and Ngam. At the time of the interview, she was looking for jobs and staying with Wan temporarily. She said she is one of the poorer people in her village. She has two children; her daughter has now come to work in Thailand while her son is ordained.

Apart from the domestic migrants, I also interviewed two employers, Ranee and Luxika; a development worker, Kanokporn and a monk who operates the merit making activities. Ranee is my relative. Luxika is the mother of my friend from high school.

#### Social Capital and Migration Process

According to the interviews, my informants are part of chain migration (Castles and Miller, 2009: 29). They are not the pioneer migrants in their communities as their movement was facilitated by the "strong and informal" social capital such as friends and relatives in the same village who have been working in Thailand before (Putnam, 2002: 10 -11). Most of them are the first generation in their family that move to Thailand.

The main reason that motivates my interviewees to come to Thailand is economic opportunities. They felt that they were poor and wanted a better life for themselves and their

family. The economic activities in their home town are working in the field, doing some small daily wage work, and selling their product. These activities only enable them to survive on a daily basis and provide them no capital surplus. Moreover, even skilled workers in Burma are paid less than domestic workers in Thailand.

"There was no work in Burma, most of the work was daily wage work that I had to do to buy rice day by day, I had no savings as there are so many people in my family"<sup>6</sup>

The migration processes though is possible because of social capital they have. Similar to the roles of social capital in assisting or influencing the migration process as described by Faist (2000: 145), for my informants, social capital is a source of information regarding the economic opportunities in Thailand. They, for example, receive information on employment, expected salary, how to move to Thailand and encouragement to migrate. Many times the networks are the one who arrange everything for potential migrants to come to Thailand.

"My relative said the salary is low here [Mae Sot – Thai border town], so she told me and my boyfriend to come to Bangkok...then my relative found me and my boyfriend a job"<sup>7</sup>

Some migrants used their social capital as a safety net guaranteeing that helped them decide to migrate. Having trusted social networks, they were ensured that they would have some support in Thailand especially when they had to move to a place with a completely strange context.

"My mother told me that since I did not have the number of pi<sup>8</sup> Moo [alias – her relative], ... there would be no one to help me if I had troubles in Thailand, but this time since p'Moo recommended (the agent) for me, I decided to come...[o]nce I arrived, she found me a job,"<sup>9</sup>

During the irregular movement process, migrants' destiny totally depends on brokers which mostly are informal and irregular one. In this case, having a social network personally recommending a place to work or migration agencies means migrants will face less risk. Their friends or relatives would know who will be responsible if migrants faced abuses

<sup>&</sup>lt;sup>6</sup> Wan, 13/02/14

<sup>&</sup>lt;sup>7</sup> Wan, 20/02/2014

<sup>&</sup>lt;sup>8</sup> It is a Thai word to refer to someone who is older, usually imply close tie.

<sup>&</sup>lt;sup>9</sup> Ngam, 16/02/2014

during the migration process. Otherwise, migrants may face problems and are not able to seek help as described by Nid.

"There were five workers [coming together with her to Bangkok], and there was one who was forced to sleep with the broker when we rested at the hotel, but what could I do?, It was totally dependent on luck at that point."<sup>10</sup>

Moreover, in the case that the trust between migrants and their ties is high such as being members of the same family, networks are also sometimes responsible for the cost of migration.

Similar to the point makes by Fasit' (2000: 140), the migration process of new comers is made possible by the reciprocal norm among the villagers that helps sustain a transnational network of support. Although, this norm is not explicitly mentioned, it could be easily assumed from the interviews that to follow this norm is vital for the villagers. It is common that villagers might receive some form of social sanction such as being a target for gossip if they refuse to help their fellow villagers. For example, when I interviewed Wan, she mentioned that people from the same villages often ask for her assistance to find employers in Thailand, but she is often reluctant to help. She chooses to give excuses to them such as lying that there are no jobs available. The reason she doesn't want to get involved, in helping others from her village, as in the quote below, explicitly shows the degree of reciprocal expectation she was subjected to by her community.

"It feels like duties and responsibility (to help new comers until they are able to adapt) [----](what will happen if she doesn't keep being responsible for the new comer who she finds a job for?) their families might be angry and mistreat me such as gossip about me",<sup>11</sup>.

It should also be noted that since Wan has worked in Thailand for a long time and has worked with her employers for almost ten years, she is already quite well adapted to Thailand. Therefore, she no longer needs as much assistance from her village's networks. Once her position in the village's network has turned from help-receiver to help-provider, she considered that it is not worthwhile anymore to continue to reinforce these norm/practices.

<sup>&</sup>lt;sup>10</sup> Nid, 22/02/2014

<sup>&</sup>lt;sup>11</sup> Wan, 13/02/2014

While the theory mentioned how social capital facilitates migration, it doesn't explore how relations with social capital stimulate these females to migrate. Considering the cultural context in the village, according to the interview, the norm of obligation toward parents wellbeing is strongly observed by villagers. Taking good care of their parent materially and mentally is an expected duty for individuals that should be strictly observed both as an admirable duty in the eye of the community and as a moral obligation. Thus to gain money in order to improve their parents and family living standard is main stimulation of all my informants to migrate. Morally, the concept of karma has been cited in some interviews as a reason to conserve the norm. If this norm is not observed then these females could encounter gossiping both for what the villagers perceive as their malpractices and for their potential unfortunate faith deal to their bad karma. For some informants, to follow this norm means an investment as they expect that their good deeds will be beneficial for them in the next life.

"I remit money both to my parents and my husband's parents ... as my in-laws cannot work anymore so I should be responsible for them as well ... I'm afraid of karma if I don't do that."<sup>12</sup>

How relationship with social network stimulate migration could also come from seeing the different between oneself and those in the same social circle. Almost all of my informants especially those who were aged around 30 -40 have emphasized that the financial success of other villagers who were working in Thailand was their main inspiration to migrate. The success is displayed through improvement of economic status such as building new big houses for families which often simulate desire of their fellow villagers to do the same.

### Social Capital and working life in Thailand

• Fellow Villagers

Similar to the process of migration, my informants also mainly rely on the strong informal tie of people from their village who work in Thailand when they live in Thailand as a domestic workers. Their social network could provide the newcomers some important skills for working in Thailand such as basic Thai words and how to clean employers' houses properly

<sup>&</sup>lt;sup>12</sup> Baitong, 18/03/2014

according to their employers' taste. Most of them mentioned that there are big differences in the perception of cleanliness and details of cleaning between their employers' space in Thailand and their home in Burma.

"Thai people's house are sooo clean that I felt I should not step on it, I have never lived like this before....ironing cloth too, when I had to do it I almost wanted to cry. Here is not like at home where ironing cloth is such a simple matter"<sup>13</sup>

Social capital also provides an economic and social support when migrants need it. Almost all my informants except Aye said that there are a lot of people who ask them to help financially. The support can also come in the form of providing shelter and a place to stay when migrants are unemployed.

"I lived in my relatives' house for two months when I tried to find work"<sup>14</sup>

Borrowing money from friends is one of the main economic supports and sources of credit among migrants. However, Bear<sup>15</sup> is the only informant that has participated in self setting rotating credit groups. She and around 15 close friends from the same village set up a saving group where they put their money together as an emergency fund for people in her own group. This group provides them financial assets when needed and helps them to rely less on borrowing money from other people which always has high interest rates. Nevertheless, this practice requires a high level of social trust. All members of the group are people who know each other well and are expected to share the same norm/obligation.

"I trusted them and there is no one who has cheated yet, if some of them cheat then no one in the village will talk to them again,"<sup>16</sup>

The networks also continue to be the most important source of information that shapes migrant's awareness and perspective, through words of mouth and informal conversations. The information that is discussed and shared could be an inquiry on the well-being of their own family, discussion on behavior and news of other migrants, consulting and encouraging

<sup>&</sup>lt;sup>13</sup> Nid, 22/02/2014

<sup>&</sup>lt;sup>14</sup> Nid, 22/02/2014

<sup>&</sup>lt;sup>15</sup> Bear, 12/03/2014

<sup>&</sup>lt;sup>16</sup> Bear, 12/03/2014

friends who have problems, giving information on places to seek social service or updating on their working experience and work related information.

Sharing information on work experience could form an expectation and raise awareness on their job condition such as the rate of salary that longtime employees should receive.

"I met a person from my village in the van back home, she told me that since I had been working here for a long time I should get paid a lot",<sup>17</sup>

Almost all informants rely on their fellow villagers when they want to find jobs or change jobs. Additionally, the person who recommends a new job themselves also found that information through their own social network or their employer's social networks. While it is easy to find a job via the "informal ties" (Putnam, 2002: 10 -11), this method of job searching often doesn't provide the negotiation power on working conditions as much as finding a job via a formal recruitment agents does. One of the reasons is that migrants often acquire jobs for their friends through the network of their employers. However, as they themselves already have an unequal relationship with the employers; it is more difficult to negotiate for proper working conditions for other migrants. Not to mention that migrants themselves are often not aware of labor rights and laws. Therefore, migrants have to take a risk on their employment condition.

The choices for employment might also be more varied with the agency service. For instance, Cherry found her current job with a foreign employer via an agent service and has a formal working contact. However, migrants must have enough life skills and be familiar with the context of Thailand enough to rely on other networks that are not their strong tie social capital. Cherry, for example, is able to speak Thai and has a lot of knowledge; thus she found an access to recruitment agents.

Migrants finding work through the assistance of an informal social network actually seems to be more beneficial to employers because they take on a lower level of risk than when they hire a stranger. For example, they find it to be easier to find a surety (the worker recommender) in cases where the workers have committed a behavior that they do considered as improper such as running away. Therefore, the migrants who recommend new workers

<sup>&</sup>lt;sup>17</sup> Nid, 22/02/2014

must find some ways to deal with the expected responsibilities from employers when they decide to help finding jobs for their social networks.

Their isolate nature of work means that these contacts with their fellow villagers have been kept very actively through mobile phones which according to the employer "are ringing all the time"<sup>18</sup>. Unlike what Putnam (2002: 16 -17) points out on the irony of technology that could increase the chance for communication but decrease actual interaction within the network; in the context of migrant domestic workers, mobile phones have enormously increased the chance of contact. The interaction within networks as mobile phones can be used to make an appointment when they have free time and to inform the network on social events such as merit making events.

Ties with their villagers also play an important role in helping migrants facing a crisis situation. For example when they want to leave abusive employers but don't know how to do so. Sometimes, in a case like this, the assistant from other people who are not migrants is vital in dealing with things that migrants from Burma aren't able to do. The example below shows that although the fellow migrants is the one who are requested for help and operate the escape plan, a "very thin" capital such as people whom they meet regularly without actual contact could also be resources in a crisis situation (Putnam, 2002: 10 -11).

"Her [the migrants who asked her for help] employers also threatened me to gave her money within an hour otherwise she would send the girl to the police, the girls cried and asked for help... since I do not even know how to transfer money I had to hire security guard to transfer money for me, I didn't even know where does this girl live so I asked her to call a taxi and told the taxi direction on mobile phone, the taxi sympathize with the story so he took her to me free of charge."<sup>19</sup>

Helping other migrants in a risky situation could also enhances one's status in the network, such as how Wan feels that Ngam is a good person as she has helped recruit the Karen migrants who were cheated by their employers.

Although this is not explicitly said in the interview, according to my observation as someone who is similar with the context of migrant workers, the assistance of the villages' network is

<sup>&</sup>lt;sup>18</sup> Ranee, 18/03/2014

<sup>&</sup>lt;sup>19</sup> Bear, 12/03/2014

crucial because there are not many other capitals available for the migrants. For example, as they came to Thailand irregularly without much education, they most often cannot use or access to the service of governments or formal recruitment agencies in order to come to work in Thailand or in order to seek help when they are mistreated. Therefore, it is not surprising that they must rely on their own informal social network throughout their working life in Thailand.

• Employer

It is not surprising that the migrant – employer relationship is filled with conflicts and many times employers can easily take advantage of domestic workers. It could be said that the nature of the relationship between employers and workers is a power relationship in a similar way to patron-client relationship where both side receive mutual benefit from the relationship but the power (and benefit?) is unequal (King, 2009: 157). However, it doesn't mean that the unequal relationship cannot co-exist with other aspects of the relationship in the perspective of migrants. While some migrants like Cherry prefers to keep the working relationship as professional as possible and try to remain independent from employers, other have a different perspective on the working relationship. Depending on different contexts, employers actually could be one of the main social capitals for migrants.

From my informants' experience, employers play a crucial role in migrants' adaptation to Thai society; they often are the one who teach migrants Thai language, work and expected norms in Thai society. The employers also assist migrants in dealing with Thai authorities such as finding schools for their children or helping migrants when they have troubles with police. They also serve as one of the main sources of economic capital for migrants when migrants need to use money. Some employers also provide food and accommodation that is better than the average of Thai society.

Due to being different in almost every aspect except for gender in the case of female employers, they can give migrants new knowledge and perspectives. They can foster their maid to the idea of individuality and future plans by giving migrants advice not to remit all money back home. For example, Ranee<sup>20</sup> told me that she is quite skeptical toward the practices of remitting all money home. She felt that the practice left migrants without capital

<sup>&</sup>lt;sup>20</sup> Ranee, 18/03/2014

for themselves to be able to save for the future. According to her, the money remitted home is sometime not carefully used. Thus, she tries to teach her maid to think of herself and her own future not only about her family.

Since employers are often older, combined with the power relation, the workers especially those in older generation often treat them as adult figures and treat their employers' instructions as those of older relatives. For example, when one of Ranee's<sup>21</sup> workers had a problem with men, her sister informed Ranee about it and asked Ranee to help supervise the worker. Similarly, as in the experience of Bear, a welcoming and kind employer can also fulfill the role of family that the migrant doesn't have in Thailand.

"I was so lucky that I had such a kind employer who understood me, she taught me everything, taught me to work, taught me to speak Thai. She took care of me almost like her children, allowed me to eat everything, took me out to travel ... [she] took care of me like her own family member and I didn't feel that I lacked anything"<sup>22</sup>

This kind of relationship can be developed to the point that migrants are considered as part of the employer's family and that they are genuinely concerned with each other's well-being. For example, Ngam who used to promised her current employers that she will take care of them all their life fulfilling the role of daughter that her employers don't have (before she decided that she wants to ordain).

This kind of relationship though could give migrants an emotional cost when the relationship with their employer is ended. For example, Ngam is concerned about how her employers will manage once she returns to Burma. Many times the relationship is kept even after the employment is ended and some employers still provide assistance for migrants when needed.

All in all, employers can actually be one of the most significant social capitals for migrants and it can be said that it is the only strong tie bridging capital that the migrants have. Also, many aspects of the relationship transcendence the notion of migrant as total victims under the power relation mentioned by King (2009: 157) above.

<sup>&</sup>lt;sup>21</sup> Ranee, 18/03/2014

<sup>&</sup>lt;sup>22</sup> Bear 12/03/2014

# • Religious Institution

Another very important bridging social capital in the working life of my informants is the temple. Almost all of them except Cherry said that they like to visit the temples. They prefer to go to the temple rather than to the department store which is a famous activity for many migrants according to my informants. The temple provides them a space that offers a peaceful feeling they cannot find elsewhere. Considering their context of work where work place and home are the same place, it is quite understandable that they would find the temple to be a place to escape to. The religious institution can also have deep effects that it shifts one life goal and how one perceives the world as in the case of Ngam whom discover the teaching about nirvana and change her life goal to eliminating her passion instead of settling in Thailand.

Religious institutions and ceremony also provide a platform for building ethnic community and reproducing collective identity (Castles and Miller, 2003: 40). According to my informants, most of the religious ceremonies are ethnic-based. This gives them a chance to reinstate their ethnic identity. The religious ceremonies, including the one I attended, always involve displaying of identity. For example, cooking ethnic food together like they have done at home, wearing ethnic dress, or listening to the sermon in one's own language. These are things that they don't have a chance to do in their work environment where they have to adapt to the space and rules of employers (Rahman, Yeoh and Huang, 2005).

• Development Organization

Of all of my informants the only two who have access to and are aware of development organizations are Moei and Cherry as I met them at the vocational school for migrants. Participating in the activities of development organizations has introduced them to new knowledge and perspectives. According to Cherry and Moei, through the school, they are informed about labor rights and health rights. They have also become more aware of the channels to seek help. This knowledge enhances their individual agency and capacity such as how Cherry feels that she is not obligated to work for employers who doesn't respect her rights<sup>23</sup>.

<sup>&</sup>lt;sup>23</sup> Cherry, 23/02/2014

The activities of development organizations also provide a chance for migrants to interact with people in different occupations as well as people with different backgrounds. This can give them new awareness and information that they can use in the future. For example, after the class in Sunday vocational school I heard some volunteers talking and exchanged with each other on the political situations in Burma. They informed each other about the road construction plan that is going to be implemented and also talked about Asean communities. Cherry and Moei themselves have thought of their future in the Asean economic community. This is something that my other respondent such as Wan has no knowledge of.

Thus, when referring to the debate on the roles of NGOs toward migrants, I'm more inclined toward the position of Ford (2004 – see background information/ literature review). The case studies of my informants have shown that NGO organizations could contribute to migrants' agency in various ways.

### Trust and cost of social capital in the context of being domestic migrants workers

In the previous section I have examined the role of social networks and norms in the lives of domestic migrant workers in Thailand. In this section, the role of trust in the migrants' social capital and the cost of keeping the social capital they have will be examined.

Apart of their employers in which the nature of works make the interaction unavoidable, as mentioned above, the networks that my informants have been in contact with the most in their daily lives are strong, informal and bonding social capital namely their fellow female villagers who are also working in Thailand. When I asked them who they would ask for help apart from their employers, almost every informant answered their friends from their village.

Although one might assume that the reason for keeping a few homogenous ties is because of their isolated nature of work, I have found out that it is actually the domestic workers' preference especially those who are Karen. Apart from Bear and Nid, the rest of my informants mentioned at least once in the interviews that they prefer to not to have too much contact with others apart from their close friends. One reason that they only keep contact with people from their village is lack of trust in other networks that some of them found to be insincere. Especially when there are no obligations or norms to secure that they will not be

taken advantage of, unlike in the relationship with people from their village whom they grew up with.

My informants, for instance, build trust via being familiar with each person's personality and knowing that there is a mechanism in place to foster obligation toward each other. They know that their friends will not be accepted in the village if they scam other villagers (Faist, 2000: 104).

According to the interviewees, the reasons for the lack of trust toward other networks that keep contact between networks minimal is various.

First, some of the migrants are taught not to trust strangers and to be scared of them. They are taught that having an interaction with a stranger means being in an unpredictable position that can be harmful since they do not know the stranger's background.

"I do not have any more friends than those whom I know from the village [refugee camp] but I'm not bored at all and am more comfortable as I don't know how other people would treat me, they might not like me and then bring me trouble."<sup>24</sup>

Therefore, the nature of domestic work is desirable for almost all my informants. Working in private households means they can avoid having problems with co-workers. They also feel that there are less negative influences from peer groups that would distract them from saving money.

"People who work in a factory don't have discipline and rules. When they have a day off they will travel and spend money. Domestic workers on another hand almost have no day off so the money is not wasted. If I have a friend who is bad influence they might make me want to dress beautifully which is a costly activity. While working as domestic worker I can just wear shorts and t-shirt."25

The non-trust can also cause by bad experiences of getting taken advantage of by people who are not sure would share a mutual obligation with. For example, when the migrant workers who they are not close with ask to borrow money from them; these types of person are the ones who most of my informants despise and want to avoid. Although sometimes avoiding

<sup>&</sup>lt;sup>24</sup> Aye, 18/02/2014 <sup>25</sup> Ngam, 16/02/2014

letting other borrow money can be difficult as the norms of kindness, generosity and manners are expected. Ngam mentions that "If they ask me for the third time and I still don't give it then it means that I have no kindness and I lie"<sup>26</sup>. This shows how having wider networks doesn't always mean benefits, if there are no collective norms, obligations and solidarity to sustain the trust.

Nonetheless, it does not mean that associating with only friends from the village will bring them fewer troubles as they would like. To associate with the people one has strong ties with also means keeping expected obligations within the network (Putnam, 1993:103). They could end up receiving negative consequences if the expected duties are not fulfilled such as Wan's description of why she does not want to find jobs for people in her network (see "fellow villagers").

Associating with the people from the village also means the ties with home communities are strong. They could be constantly monitored by their peers who serve as a transnational linkage transmitting information on their behavior between their home community and Thailand. Thus, the migrants still have to act in a way that conforms to the norms of their communities even if they are currently living in Thailand. It can be seen that actually transnational links provide a way for local communities to influence migrants' lives in another country as well. Thus, sometimes choosing to alienate themselves from the social capital whom they are not truly close to has also been a choice of some workers in order to avoid the negative consequences.

### Gender, Ethnicity and Class

The reasons that my informants choose to associate with an extremely bonding social capital and not really build trust with other ties are also shaped by the context of gender, ethnicity and class. The stratification also shaped their interaction with different social capitals. As the example of my informants shows, they prefer to keep contact with people of the same gender, class and ethnicity. This section will examines how gender, ethnicity and class shape how the migrants build social trust and social networks.

• Gender

<sup>&</sup>lt;sup>26</sup> Ngam, 16/02/2014

"I was scared of Thailand at that time, many people told me that one will lose virginity in Thailand, many people I knew started their family shortly after they came to Thailand",<sup>27</sup>

The quote of Wan above shows how the belief that women belong to and are safer in private space is prevailing. (King, 2008, 198).

Considering that they are taught not to trust strangers, it is not surprising that some of the workers especially those who are still single also feel uncomfortable with men and are taught by their family to distrust men. Wan, for example, told me that the benefit of having a boyfriend is to protect her from unwanted attention from other migrants especially when she was younger<sup>28</sup>. Due to these reasons, some of my informants prefer to stay at their employer's house rather than working in a factory.

Even it is obvious that several migrant's families are concerned with their daughter's potential sexual behavior or that they are influence by the dichotomy of "Madonna and whores" (Harrison, 1999:168). It doesn't always mean that they would like to control their daughter's behavior but can be that they are concerned about whether their daughter will be able to handle being in relationship and that their daughter might get exploited. "She is still naïve and young, if she has boyfriends there are potential problems... [w]ill she [her daughter] be able to handle it [the problems of relationship]...having boyfriend when she is not mature enough might be because of lust not love<sup>,29</sup>.

Also for many migrants being in relationships brings them burdens. Sometime men are demanding and do not give them personal space. Some of the informants also feel that having too much attention to relationship would disorient their goal to find money. The quote below perfectly describes how many of my informants feel.

"There are a lot of people in factories, it could bring a hassle, I prefer to live quietly ... being a domestic worker I can do the work on my own while in a factory people could cause me

Wan, 20/02/2014 <sup>28</sup> Wan, 20/02/2014

<sup>&</sup>lt;sup>29</sup> Mom. 09/03/2014

troubles, I also don't want men to court me, as men in factories are often already married and I don't want to have a boyfriend right now as my goal is to make money".<sup>30</sup>

Not to mention that they might get tricked monetary or used as sexual partner for men who already have a family.  $^{31}$ 

As mentioned before, uncontrolled sexual behavior could also cost their reputation in the village. Therefore, the context of being domestic workers living in private households is used as a space to avoid potential problems. Nevertheless, it should be noted that although my informants highlight the negativity of associating with men, it doesn't means that they completely controlled themselves sexually in practice. Ranee told me that dealing with the love problems of her maid is one of the things that gives her the most headaches (she describes a lot of relation problems of my informants that she knows but I will not address those information due to ethical consideration). Therefore, it is possible that I am informed about proper sexual behavior because it is considered to be a good image to present rather than the actual practices that my informants follow.

Resonating with the literature on transnational mother/daughter (Ueno, 2013; Thai, 2012; Boccagni, 2013 and Parrenas,2011), according to some of my informants such as Ngam<sup>32</sup> or Bear<sup>33</sup>, it seems that female migrants also are more likely than males to keep the obligations towards their parents while living in Thailand. Ngam told me that "[Men] start to behave badly once they come to Thailand, women also save money for family, but men when they go back home sometimes don't even have savings...there are only five from hundreds that still behave properly". This could be another reason that kept my informants from associating with the opposite sex whose characteristics according to their gender perception could detract them from intended goal.

• Ethnicity

Ethnicity is very important to the informants as all of them identify themselves by ethnicity first rather than by nationality. Even with fellow workers from Burma there is less feeling of solidarity and trust between people from different ethnic groups, not to mention about the

<sup>&</sup>lt;sup>30</sup> Apple, 08/03/2014

<sup>&</sup>lt;sup>31</sup> Ranee, 18/03/2014 I got this information from employers not migrants.

<sup>&</sup>lt;sup>32</sup> Ngam, 16/02/2014

<sup>&</sup>lt;sup>33</sup> Bear, 12/03/2014

Thai whose different statuses cause the power relations in the first place. When I explained the purpose of the interview to Wan that I would like to study Burmese (as nationality) domestic workers in Thailand, she told me right away "but I'm not ethnic Burmese",<sup>34</sup>.

The words as "both of us are Karen", for example, have often been cited as a reason for some of my Karen informants when they choose to interact with someone including when helping those in need or when they choose to associate with or start a conversation with someone.

As I have found in my field research, the consciousness on ethnic differences and less association between different ethnicities could be either because of small reasons such as the language differences or it could be a result of assumptions towards norm and characteristic of another ethnic group. Ngam told me "Most of Karen are good people, for Mon they do not know how to save money, use all their (daily) salary everyday".<sup>35</sup>

According to my observation, the sense of ethnic's belonging transcends nationality and border. At the Karen merit making activities I have observed, the monk who takes charge of organizing the activities and fundraising is actually a Thai-Karen monk not a monk from Burma. But he is view as one of the Karen and is trusted by migrants who have attended the ceremony. The ceremony I have attended is actually a merit making ceremony for his birthday. This seems to be a contrast situation to the interview of Wan where she states that her husband who is ethnic Burmese always feels out of place in Karen companies<sup>36</sup>.

It also shows that ethnicity, according to the informant, is a mechanism for building social trust as well.

"Since I speak the same language as them, they lets me be the one who organize [the activities and managing the fund]"<sup>37</sup>

• Class

It seems that for my informants, class is also as important as ethnicity in defining social relationship. For example, sometimes they feel that they are in the position of being looked

<sup>&</sup>lt;sup>34</sup> Wan, 09/02/2014

<sup>&</sup>lt;sup>35</sup> Ngam, 16/02/2014

<sup>&</sup>lt;sup>36</sup> Wan, 20/02/2014

<sup>&</sup>lt;sup>37</sup> Monk, 28/03/2014

down on by other people in Burma who have better social status. "Some people who are so high educated like to look down on others",<sup>38</sup>. Also, some of the informants such as Bear and Nid have developed close ties with people of different ethnicities but belong to the same class with them.

The notion of class could be embed into individual thought and controls their social action which could affect migrants' relation to social capital. The case of Wan's youngest daughter is a good example. According to my observation, due to her employer's support, Wan's daughter was raised with a Thai middle class's life style. She told me weary that when she took her daughter to swimming pool, people think she is the care taker of her daughter instead of her mother due to her daughter's action and characteristic. She seems to be uncomfortable with it.

How Wan feels that her daughter's middle class life style is not proper to their status provides an example of why it might be difficult to develop a more equal relationship of between people of different class in term of norm and obligation. As the feeling class shapes and limits the possibility of one's social action.

According to Wan, for example, having a "strong tie" between people of different class seems to be unimaginable (Putnam, 2002: 10 -11). When I asked her why she doesn't try to marry someone who is not low skilled labor, she told me "There is no way that kind of people will be interest in us and even if they are interested I don't think I will agree anyway because it is weird"<sup>39</sup>.

The notion of class difference is deeply integrated to the working relationship in Thai society. In Thailand, it is common for domestic workers (and other occupation as well) to refer to the employer as "master".<sup>40</sup> Because domestic worker is an occupation that is treated with lowest social status in Thailand, the awareness of class different and social status is even more solidified for some migrants due to the ways they are treated (Toyota, 2005: 290). Such as Meoi who felt that she was treated the lowest that she has ever been treated in her life, as a different type of human being than her employers. "I had never thought that I had to eat

<sup>&</sup>lt;sup>38</sup> Cherry, 23/02/2014

<sup>&</sup>lt;sup>39</sup> Wan, 18/03/2014

<sup>&</sup>lt;sup>40</sup> เจ้านาย

leftover food<sup>41</sup>, if someone used to starve then I don't think they will mind, but I have never starved before".<sup>42</sup>

This awareness, however, causes Meoi to think of social mobility and change her life goals from having a lot of money into having respect from others.

"I want to go back because I feel it is useless being here, if I continue working in Thailand I will be subservient for all my life",<sup>43</sup>

• Solidarity and Social Capital as Public Good

Although social categorization has shaped and limited migrants' relationship with social capital, it is also the main source that fosters the feeling of solidarity. This feeling brings out the "public good" characteristics of the migrants' social capital. This section will also explore the benefit of social capital that migrants possess in the public level.

The awareness of being different both in term of class and ethnicity including by being mistreated can produce a feeling of solidarity within each "symbolic tie" (Faist, 2000: 102; 109) . Faist (2000: 109) pointed out that solidarity can be built through "a willingness to transcend immediate self-interest, grounded in emotional identification with others." Due to their experience in Thailand, some of my informants have developed the feeling of sympathy for other people who belong to the same "symbolic ties" (Faist, 2000: 102; 109). For instance, being treated badly because of her social status inspires Cherry to educate people from Burma. Education, she hopes, will provide her peers knowledge to be able to protect themselves from the same treatments she received in the past<sup>44</sup>.

Creating the dichotomy of "us and them" though could be the cause that solidifies the connection with symbolic ties (Richardson, 2007:51). "Them" in the cases of my informants are often Thai people or Thai employers whom some such as Cherry and Meoi perceive as oppressors of well being of migrants from Burma. However, this also means, once dichotomy is created, it is more difficult to foster trust between groups within the dichotomy as well.

<sup>&</sup>lt;sup>41</sup> It is a practice in many employers' home that they will eat first and domestic workers will eat later the food that the employer's family doesn't finish

<sup>&</sup>lt;sup>42</sup> Meoi, 23/02/2014

<sup>&</sup>lt;sup>43</sup> Meoi, 23/02/2014

<sup>44</sup> Cherry, 23/02/2014

"After that, I never want to associate with Thai people [She means Thai ethnic not Thai nationality] again, I hate Thai and always beware that Thai will not treat Burmese well. I hate so much, I felt so hurtful." (Moei, 23/02/2014).

This quote from Moei is also very interesting as, in contrast to the finding on ethnicity above, she uses the word Burmese to address migrant workers rather than using her own ethnicity of Karen. It could be assumed that she is speaking in the context of Thailand where people from Burma are treated as one entity both by the state and employers and this also influences how migrants identify themselves as well.

Putnam states that weak ties are more important in "sustaining community cohesion and collective action" which increases a platform for public benefit (1993: 175 - 176). From my field work, this point is demonstrated through the merit making and fund raising for Monk education activities that I have attended. In this activity the participants are united through symbolic ties of being Karen from the same districts that want to contribute to their homeland rather than strong personal ties.

The vocational school which gives an education opportunity for people who are linked by very weak ties of having the same nationality is also a good example. Both of these capitals show how collective actions can contribute to the people who have not directly taken part in maintaining and accumulating capital. For example, the Karen people back in the village who receive the education from monks.

Even though the activities such as merit making or vocational school provide public benefit and brings various people together, in the end, it is still shaped and driven by social status. The activities of these two social capitals are involving specific groups of people that are tied together by pronominal attachments toward to Burma or to a certain ethnic group. Therefore still doesn't serve as a linkage of different social group between those of migrant workers and other groups in Thailand, and has not enhanced norms and trust in the wider level.

## Social capital and migrant's agency

Although there are capitals available, agency is crucial in determining how the capital is used (Charrad, 2010 ;Ling and Dale, 2013: 4-5). As individuals are linked to the wider society, in order to achieve their goal, they must shape and change their social network as well (Ling and Dale, 2013: 5). From the interviews with my informants, agency is evidently displayed.

While social capital is important in facilitating migration, it requires migrants' strong will in order for migration to take place. Coming to Thailand for many of my informants meant they entered into a completely strange place that they had almost no knowledge of. As illustrated above, even if there is assistance from social networks, the fact that they cannot speak the language, don't know channels to get help and must prepare for an unpredictable risk means that the decision to migrate requires tremendous agency.

Although it could be argued that reciprocal norms and obligations towards family place burden towards migrants and cause them to accept all the risks relating to the migration process, I would say that, for my informants, contributing to family is their personal goal and investment. But the benefit from this investment might be different than how benefit is conceptualized from the perspective of capital. For example, as mentioned before, some informants conceptualize their obligations as a chance for merit making that they expect to yield them a better life in the future (or next life).

The feeling of obligation is not only derived from norms. Wan, for example, wants to make her mother live comfortably because she is appreciative of how much her mother has sacrificed for the family.

"My mother has always lived in hardship, I want to give her when I still be able to  $\dots$ my mother is the most important person"<sup>45</sup>

Not to mention that, for some of the interviewees, a personal agency of wanting to have new experiences, curiosity or even to display challenges to gender notions within their network also contributed to the migration.

"My mother doesn't love me, she only loves her sons.....when I was at home she always complained about me that my ear got numb, so I think it is better to leave and prove to her that I could do like her son"<sup>46</sup>

The clearest example of displaying agency is how migrants deal with abusive employers. If they are treated badly or are not paid properly, most of my informants are not passive in the situation. Although none of my informants have used legal mechanisms to address their

<sup>45</sup> Wan, 09/02/2014

<sup>&</sup>lt;sup>46</sup> Bear, 12/03/2014

abusive employers and often use the choice of running away, the decision to run away especially for irregular migrants requires a lot of capability. They must decide, calculate and plan in detail on the process of running away that will enable them to leave and reach their support network without being caught. Nevertheless, the most important factors are the boldness and readiness to face obstacles in order to make change. This feeling sometimes makes them determined to leave when even a small chance is present.

"They ordered me to buy stuff in markets, so I used that money for transportation cost to my sister's place, I left with nothing",<sup>47</sup>

Although they live under their employers' rules and space, most of my informants don't take employers' treatment or working conditions for granted and often negotiate. For example, they can negotiate for a rise in salary. If the workers have worked for a long time and are trusted by employers then they might use a tactic of threatening to leave in order to increase their negotiation power.

As I mentioned above, the cost of having social capital such as being the target of gossip or having to follow expected behaviors can be an annoyance and burden for migrants. For example, when someone would like to borrow money from them. Therefore, various informants have used several strategies in order to avoid the obligation. The strategies could be lying that they don't have money such as Baitong, or avoiding having too many new friends such as Wan. But Ngam has a unique practice as she uses religion to deal with the situation. "It might be because of my karma that the money is not returned … if I can get it back that would be good, if I can't then it is a chance to make merit".

While some might assume that Ngam lets other people take advantage of her and that the religious teachings sustain this kind of practices, I would argue that Ngam does not passively let someone take advantage of her and uses the religious teaching to legitimize the passive practices. Although she believes in Karma, she still tries to get her money back in various ways. Before she lets someone borrow her money, she has already thought and calculated thoroughly on the consequences of her actions. Religious teachings have actually been used to give her peace of mind on the consequences that she cannot control. This is a way she uses

<sup>&</sup>lt;sup>47</sup> Bear, 12/03/2014

<sup>&</sup>lt;sup>48</sup> Ngam, 16/02/2014

to deal with the negative part of having networks. This shows her own way of exercising resistance.

Interaction with social networks in the context of Thailand could also enhance migrant agency. The awareness of their status from being treated badly and being in a disadvantage position under the social structure is an inspiration to improve themselves. Even under limited circumstances and structures that doesn't give them many opportunities, some of my informants constantly try to find ways to invest in a better future. This could be in the form of improving themselves through accumulating knowledge and skills as in the case of Moei and Cherry.

"I realized that my life has no choices because I have no education, no education means I only can work in a subordinate position"49

"I want to find new experiences for my future, if I have more knowledge then it is easier to find a job."<sup>50</sup>

Sunday is often the only day off for most migrants. The fact that they choose to come to school instead of having leisure time shows how the agency is exercised to suit their goal even in limited circumstance.

Migration experience in Thailand, on the other hand, has shaped Ngam's goal and heightened her agency to achieve it. Life as a migrant has made her understand the uncertainty of life and feel the need to embrace the nirvana which she sees as the best solution for her life. Wanting to achieve this goal, she thinks of going back to Burma as soon as she can in order to become ordained. Although she has received a lot of disagreements from her social network, such as from her husband, her agency is strongly exercised to the point that her social network cannot influence her decision to pursue it. She has negotiated and attempted in various ways to alter the network's attitude. For example, she brings her husband to learn religious teachings in the hope that he will perceive the world in the same way as her.

"[M]y husband doesn't want to go back but he can't live here alone....so I reminded my husband that we have to part one day when I died anyway",<sup>51</sup>

<sup>49</sup> Moei, 23/02/2014 <sup>50</sup> Cherry, 23/02/2014

All in all, it could be assume that migrant agency is the driving force throughout the whole migration process including in the relationship of the migrants and their social network. Without agency, the female would not decide to migrate in the first place. From this perspective, it seems that social capitals and migrant's agency mutually shape one another. Social capitals are used to accommodate the personal goal of the migrants. On the other hand, migrants' personal goal is also shaped by their social relations.

# **Chapter 5: Concluding Remarks**

The aim of the thesis has been to investigate the meaning of social capital in the life of domestic migrant workers from Burma in Thailand. The first and second sub-research questions attempt to see the role of social capital in migration process and their life in Thailand. The results show that social capital is involved in every stage of the migration process and the informants heavily rely on their social capital to be able to live and work in Thailand. These networks they have is tied by trust, norm and social obligation (Putnam, 1993; Putnam, 2002).

However, having social capital also means that the migrants have to accept what the social capital costs and that includes the importance of following norms and expected obligation. Therefore, I agree with Planthaloet (2012) that each network also could be a burden or a source of conflict that constrains migrants as individuals depends on the context and depending on the relationship with social networks. This side of social capital has not been much emphasized in the literature on migrants' social networks.

The type of social capital they have also defines how the capital is used and the effects of social capital toward the migrants' lives in Thailand.

According to my informants, the social capitals they keep and invest in are extremely bounding (Putnam, 2002: 11). Most of them only have a close association and have a chance to associate with people of the same ethnicity, class, occupation, gender, and place of birth. Also, similarly to the finding of many literatures on low-skilled transnational migrants

<sup>&</sup>lt;sup>51</sup> Ngam, 16/02/2014

(Nurchayati, 2011; Johnson, 2011 ; Ueno, 2013 ; Kerdmongkol,2011), the informants rely more on the transnational ties they have with the people from the same village back home rather than new networks in the host country. Apart from their employers, the only other types of social capital in their lives that could be considered as bridging capital are religious institutions and development organizations. However, the activities of these organizations are still created from pronominal attachments of belonging to a certain ethnicity or nation (Castles and Miller, 2009: 35). In relation to the point above, the thesis also found out that migrants rely mainly on informal social capital rather than formal mainly due to inaccessibility. I would say that the only formal social capital they access is development organizations.

All the factors above have limited migrants' choices, options and access to social capital and don't give them much chance to have capitals that provide a benefit that is more suitable to their goal. The reasons for this could be both because of migrant's personal choices due to distrust of others or because of context that provides them less access to other kinds of capital. This also means, in the context of the migrant domestic workers' lives, there seems to be less social trust, civil engagement and cooperation between different gender classes and ethnic groups both within the social circle of the migrants themselves and between the migrants and Thai society (Putnam, 1993). Therefore, it is not surprising that migrant communities are not included in Thailand migration policies-making process (Archavanitkul and Hall, 2011: 67).

In the context of the informants' lives, the thesis also found out that the weaker ties such as people at education institutions could also provide a platform for migrants to exchange and receive new ideas and be a better platform for developing public benefit; unlike people that have strong ties with migrants who often come from the same background. Thus, the thesis agrees with the assessment that "weak ties may also be better for knitting a society together and for building broad norms of generalized reciprocity" (Putnam,2002: 11)

There is not so much information according to the interview on inward looking and outward looking social capital. Apart from the fact that the only formal and bridging capitals migrants can access are outward looking organizations. Therefore, the thesis doesn't agree with the stand that migrants' social organization only should be built from below with purpose to advance for political rights that is mentioned in Ally (2005). The thesis' findings support the

notion that every relationship of migrant workers could provide them useful capital (Putnam, 2002: 6).

It should also be noted that even though their option and access to social capital is still limited, the context of migration in Thailand actually provides a chance for migrants from Burma to use various capitals that might not be available at home since life at their villages only offers minimal networks and social circles to interact with. The thesis also identify that life in Thailand can provide the context that shapes migrants' goal, choices and agency.

The third and fourth research sub questions aim to investigate the migrants' relationship with their social capital on a micro level. The third question aims to understand how the migrants' relationships with their social capital are shaped by multi-dimension social label, while the fourth question focuses on how the migrants' agency is exercised in their social relations.

The findings of the thesis supports the notion that specific context is crucial when subjects are studied (Charrad, 2010; Brittle, 2000; Herr, 2014; Nash, 2008). As shown in the thesis, gender, ethnicity, class and religion contribute to migrants' limited options and choices of social capital. But, at the same time, they can also be the platform that shapes individual's agency and nurtures feelings of solidarity. The feelings of solidarity could contribute to widen the individual's trust and feeling of obligation toward people with weaker ties but shared symbolic identity (Faist, 2000 : 101 - 102).

Considering cultural context could also highlight the possibilities of inter-subjective interpretation of the concept of social capital. For example, could the practice of using social capital as a merit making vehicle for the next life be considered as a benefit of social capital? This seems to be the case in the perspective of some migrants.

Regarding agency, the findings agree with the literature on migrants' agency that the informants are agents who can accumulate benefit from their migration experience to foster their personal goals rather than being passive victims of socio-economic structure (Nurchayati,2011; Ueno, 2013; Johnson, 2011 and Paul, 2009). It is shown in the thesis that even under limited circumstances they still attempt to manage to use social capital for their own self-advancement, and to resist constrains.

Migrants' agency and capability be could even more strongly exercised if they were provided with more chances of access to more types of social capital and social service. The thesis content reveals that personal social networks might not be the best benefits providers for migrants even if they are the resources that most migrants rely on. Actually, other institutions such as government agents or business agents might be more effective in similar roles. And, as mentioned above, the fact that they almost totally rely on personal social networks is because of a lack of access to and unawareness of other services providers. Therefore, the thesis would like to emphasize that organizations and institutions should reach out and provide migrants more chances to develop ties and build trust. The thesis hopes that this practice will not only give the benefit to migrants but could also foster social trust on a wider level. The ties should not be limited only to social service providers but formal business organizations and state as well.

I would like to emphasize on the role of the state that can effectively improve domestic migrant workers' statuses. It could be said that the migrants use their personal ties in order to compensate for lack of statement involvement in providing them security and opportunities such as non-effective labor's protection mechanism. What kind of interactions between different social capitals and what kind of norms could lead to more involvement of the state in the migrant domestic workers' welfare is an interesting question that could be researched in the future.

Lastly, I would like to end my thesis with a question toward how the concept of social capital is framed. Interviewing many informants, I also start to question the notion that a person is willing to contribute or help others only because they expect long-run benefits in return. Or is it the case that one needs feelings of solidarity in order to decide to help others as illustrated in Putnam (1993, 2002) and Faist (2000). When I asked my informants for an interview, most of them agreed to it even though it was very unlikely that they would receive anything in return in the short or long distance future. Agreeing to an interview with me didn't guarantee any reciprocity between us and seemed to cost them an extra time off their work. When I told Bear that I was sorry to waste her time, she responded: "I want to help someone who has a dream to achieve that dream". Might this act of humanity, consideration and sympathy be the unexplored mechanism of social capital?

### Reference List

Abrams, Lynn (2010), Oral History Theory. Routledge: Oxon.

Anderson, Bridget (2010) "Mobilizing migrants, making citizens: migrant domestic workers as political agents" *Ethnic and Racial Studies* Vol.33 No.1 pp. 60 -74

Ally, Shireen (2005), "Caring about Care Workers: Organising in the Female Shadow of Globalisation" in *LABOUR, Capital and Society* Vol. 38, pp. 186 – 207.

Archavanitkul, Kritaya and Hall, Andy (2011) "Migrant Workers and Human Rights in a Thai Context" in Huguet, Jerrold W. and Chamratrithirong, Aphichat (ed.) *Thailand Migration Report 2011*. Bangkok: International Organization for Migration.

Behar, Ruth (1990), "Rage and Redemption: Reading the Life Story of a Mexican Marketing Woman" in *Feminist Studies*, Vol. 16, No. 2, pp. 223-258

Boccagni, Paolo (2013) "Caring about migrant care workers: From private obligations to transnational social welfare?" *Critical Social Policy*, Vol. 34 pp. 221 -240

Brettel, Calorine B. (2000), "Theorizing Migration in Anthropology" in Brettel, Calorine B. and Hollifield, James F., eds. *Migration Theory: Talking Across Disciplines*. New York, NY: Routledge.

Bryman, Alan (2004), *Social Research Methods* (2<sup>nd</sup> ed.). Oxford: Oxford University Press. "------"(2008), *Social Research Methods* (3<sup>rd</sup> ed.). Oxford: Oxford University Press. "------"(2012), *Social Research Methods* (4<sup>th</sup> ed.). Oxford: Oxford University Press

Cambell (2012), "Cross-ethnic Labour Solidarities among

Myanmar Workers in Thailand" in *Journal of Social Issues in Southeast Asia* Vol. 27, No. 2, pp. 260 -284.

Castles, Stephen and Miller, Mark J. (2003), *The Age of Migration: International Population Movement in the Modern World.* New York: Palgrave.

Centre for East and South-East Asian Studies, Lund University (2013) Guidelines for the

*Master's Thesis*. [pdf] Centre for East and South-East Asian Studies, Lund University. [Internal Circulating].

Chamlee-Wright, "Social capital as collective narratives and post-disaster community recovery" *The Sociological Review*, Vol. 59, pp. 266 -282.

Charrad, Mounira M. (2010) "Women's agency across cultures: Conceptualizing strengths and boundaries" *Women's Studies International Forum* Vol. 33.

Chin, Christine B. N. (2005) "Neither at Work nor at Home: Asian Transnational Domestic Workers in Malaysia" in " in Huang, Shirlena, Yeoh, Brenda S. A. and Rahman, Noor Abdul (eds.) Asian Women as Transnational Domestic Workers in Asia. Singapore: Marshall Cavendish, 262-287.

Coleman, Jenny (2009) "An introduction to feminisms in a postfeminist age" Women's Studies Journal, Vol. 23, No. 2, pp. 3 -13.

Creswell, John W. (2007) *Qualitative Inquiry & Research Design: Choosing Among Five Approaches*. Thousand Oaks: SAGE Publications.

Faist, Thomas (2000) *The Volume and Dynamics of International Migration and Transnational Social Spaces*. Oxford: Oxford University Press.

Ford, Michelle (2004) "Organizing the Unorganizable: Unions, NGOs, and Indonesian

Migrant Labour" International Migration Vol.42 No. 5, pp. 99 -118.

Flyvbjerg, Ben (2006) "Five Misunderstandings About Case-Study Research" in *Qualitative Inquiry* Vol. 12 No. 2, pp: 219-245

Hall, Andy (2011) "Migration and Thailand: Policy, Perspectives and Challenges" in Huguet, Jerrold W. and Chamratrithirong, Aphichat (ed.) *Thailand Migration Report 2011*. Bangkok: International Organization for Migration.

Harrison, Rachel (1999) "The Madonna and the Whore: Self/"Other" Tensions in the Characterization of the Prostitute by Thai Female Authors" in Jackson, Peter A. and Cook, Nerida M. eds. *Gender and Sexualities in Modern Thailand*. (Chiang Mai: Silkworms).

Herr, Ranjoo Seodu (2014) "Reclaiming Third World Feminism: Or Why Transnational Feminism Needs Third World Feminism" *Meridians: Feminism, Race, Transnationalism.* Vol. 12, No.1, pp. 1 -30.

Hew, Cheng Sim (2007), "Urbanization in Sarawak: A Context" in Hew, Cheng Sim (ed,) (2007). *Village Mothers City Daughters: Women and Urbanization in Sarwak*, Singapore: ISEAS.

Hopkins, Nick (2011) "Religion and Social Capital: Identity Matters" *Journal of Community* & *Applied Social Psychology*. Vol. 2, pp. 528 -540.

International Organization for Migration (2013), "Migrant Information Note, Issue 21 – September 2013" <u>http://th.iom.int/index.php/migration-resources/facilitating-</u> <u>migration/migration-information-notes/Migration-Information-Note-21-SEP13-ENG/</u> [online] access 12th February 2014.

International Organization for Migration (2013B), "Migrant Information Note, Issue 18 – January 2013" http://th.iom.int/index.php/migration-resources/facilitating-migration/migration-information-notes/Migration-Information-Note-18-JAN13-TH/ [online] access 12th March 2013.

International Organization for Migration (2014), "Key Migration Terms [online]" <u>http://www.iom.int/cms/en/sites/iom/home/about-migration/key-migration-terms-</u><u>1.html#irregular-migrant</u>, access 16 January 2014.

Johnson, Mark (2011) "Freelancing in the Kingdom: Filipino Migrant Domestic Workers Crafting Agency in Saudi Arabia" *Asian and Pacific Migration Journal*, Vol. 20, Nos.3-4, pp. 459 -478.

Jovchelovitch, Sandra and Bauer, Martin W. (2000) "Narrative Interview" in Bauer, Martin W. and Gaskell, George ed. *Qualitative Researching with Text, Image and Sound*. SAGE Publications: Thousand Oaks, CA.

Kerdmongkol, Adisorn (2012), "Pa – O migrant workers in Bangkok: Power and Negotiation in Transnational Life" วิทยานิพนธ์ระดับบัณฑิตศึกษา คณะสังคมวิทยาและมานุษยวิทยา มหาวิทยาลัยธรรมศาสตร์.

Kim, Harris H. (2013), "Transnational ethnic networks and the creation of immigrant social capital: a multilevel analysis" in *The Social Science Journal* Vol. 50: pp. 349–358

King, Victor T. (2008) *The Sociology of Southeast Asia: Transformations in a Developing Region*. Copenhagen: NIAS.

Koetsawang, Pim (2001) In search of sunlight: Burmese migrant workers in Thailand Bangkok: Orchid Press.

Levitt, Peggy. (1998) 'Social remittances: migration-driven local-level forms of cultural diffusion', *International Migration Review*, Vol. 32. PP. 926-948.

Levitt, Peggy and Lamba-Nieves, Deepak (2002) "Social Remittances Revisited," *Journal of Ethnic and Migration Studies*. Vol. 37, pp. 1-22

Lutz, Helma (2011) *The New Maids: Transnational Women and the Care Economy* London, Zed Books.

Mahoney, Dan (2007) "Constructing Reflexive Fieldwork Relationships: Narrating My Collaborative Storytelling Methodology" in *Qualitative Inquiry* Vol. 13 No. 4, pp: 573-594.

Matichon (2007) "นายจ้างโปรดทราบ! ก.แรงงานออกกฎเพิ่มสิทธิกนรับใช้ในบ้าน ต้องมีวันหยุดประจำสัปดาห์-พักร้อนได้6วัน/ปี"

http://www.matichon.co.th/news\_detail.php?newsid=1352735089&grpid=01&catid=&subcat id= [online] access 12 Febuary 2014

Mon, Myat (2010) "Burmese labour migration into Thailand: governance of migration and labour rights" *Journal of the Asia Pacific Economy* Vol. 15, No. 1, pp 33–44.

Moses, Jonathon W. and Torbjörn L. Knutsen (2007) Ways of Knowing, Competing Methodologies in Social and Political Research. Houndmills: Palgrave Macmillan.

Nash, Jennifer C. (2008) "Re-thinking intersectionality" Feminist Review, No. 89, pp. 1-15.

Nurchayati (2011) "Bringing Agency Back In: Indonesian Migrant Domestic Workers in Saudi Arabia" *Asian and Pacific Migration Journal*, Vol. 20, Nos.3-4, pp 479 - 502.

Panam ,Awatsaya Zaw,Khaing Mar Kyaw Caouette,Therese Punpuing, Sureeporn (2004) *Migrant Domestic Workers : from Burma to Thailand* Nakhonpathom: Institute for Population and Social Research.

Panthaloet, Wannaporn (2012) "Lao Women Working in Karaoke Bar in Thailand: migration and transnational relations" วิทยานิพนธ์ระดับบัณฑิตศึกษา คณะสังคมวิทยาและมานุษยวิทยา มหาวิทยาลัยธรรมศาสตร์.

Parrenas, Rhacel Salazar (2001) "Mothering from a Distance: Emotions, Gender, and Intergenerational Relations in Filipino Transnational Families" in *Feminist Studies*. Vol. 27, Issue 2, p361-390.

Paul, Anju Mary (2009) Conference Papers -- American Sociological Association. 2009 Annual Meeting 20p. Pearson and Kusakabe (2012) "Who Cares? Gender, Reproduction, and

Care Chains of Burmese Migrant workers in Thailand" in Feminist Economics Vol. 18 No. 2.

Pessar (1995) "On the Homefront and in the Workplace: Integrating Immigrant Women into Feminist Discourse" Anthropological Quarterly, Vol. 68, No. 1, pp. 37 -47.

Piper, Nicola (2008) "Feminisation of Migration and the Social Dimensions of Development: the Asian case" *Third World Quarterly*, Vol. 29, No. 7, pp. 1287–1303

Putnam, Robert D. (1993) *Making Democracy Work: Civic Traditions in Modern Italy*.Princeton: Princeton University Press.

Putnam, Robert D. (1995) "Bowling Alone: America's Declining Social Capital" *Journal of Democracy*. Vol. 6, pp. 65 – 78.

"-----" (2002) .Democracies in Flux : The Evolution of Social Capital in Contemporary Society New York : Oxford University Press.

Rahman, Noor Abdul, Yeoh, Brenda S. A. and Huang, Shirlena (2005) "Dignity Overdue: Transnational Domestic Workers in Singapore" in Huang, Shirlena, Yeoh, Brenda S. A. and Rahman, Noor Abdul (eds.) *Asian Women as Transnational Domestic Workers in Asia*. Singapore: Marshall Cavendish, 233-261.

Resurreccion, Bernadette (2010) 'Gender Trends in Migration and Employment in Southeast Asia', in Devasahayam, Teresa (ed.), *Gender trends in Southeast Asia*, Singapore: ISEAS.

Thailand Law Center (2010) "พรบ.ป้องกันและปราบปรามการค้ามนุษย์ พ.ศ.2551"[Online] <u>http://www.thailandlawyercenter.com/index.php?lay=show&ac=article&Id=538974579&Nty</u> <u>pe=19</u> : accessed 21st May 2014. Thai, Hung Cam (2012) 'The dual roles of transnational daughters and transnational wives: monetary intentions, expectations and dilemmas' *Global Networks* Vol. 12, pp. 216 -232.

Toyota, Mika (2005) 'Burmese' Housemaids as Undocumented Workers in Thailand" in Huang, Shirlena, Yeoh, Brenda S. A. and Rahman, Noor Abdul (eds.) *Asian Women as Transnational Domestic Workers in Asia*. Singapore: Marshall Cavendish, 288-304.

Ueno, Kayoko (2013) "Love Gain: The transformation of Intimacy among Foreign Domestic Workers in Singapore" in *SOJOURN: Journal of Social Issues in Southeast Asia*. Vol 28, NO. 1, pp. 36-63.

Vaismoradi, Mojtaba (2013) 'Content analysis and thematic analysis: Implications for

conducting a qualitative descriptive study' in *Nursing and Health Sciences*. Vol 15, pp. 398–405

Wai, Myint (2004) in Bhumiprabhas, Subhatra and Kerdmongkol Adisorn, ed. *A memoir of Burmese workers : from slaves labour to illegal migrant workers*, Bangkok : Thai Action Committee for Democracy in Burma.

# Appendix 1 : the Informants' information.

- All interviewees from this section are female
- All of them work as domestic workers right now
- All of them work in Bangkok Metropolitan Area
- Except for Aye (not sure less than 5 years) and Nam (around 8 years), all interviewees have been working in Thailand for more than ten years.

No.	Name	Date	Ethnicity	Age	Place of Origin
	(alias)				
1.	Wan	13/02/2014	Karen	40	A village near Hpa-An, Karen State
		20/02/2014			
		18/03/2014			
2.	Ngam	16/02/2014	Karen	38	A village near Hpa-An, Karen State
3.	Aye	18/02/2014	Karen	19	Mae La Refugee Camp
4.	Nid	22/02/2014	Burmese	38	A village near Bago
5.	Moei	23/02/2014	Karen	24	Katom, A village between Ppa-An
					and Mawlamyine
6.	Cherry	23/02/2014	Karen	35	A district in Hpa-An province three
					hours from Pa-An city
7.	Nam	28/02/2014	Karen	40	Lai Puay , Hpa-An
8.	Bear	02/03/2014	Pa O	29	District in Mon state
		12/03.2014			
9.	Apple	08/03/2014	Pa O	29	Patom Province, 7 hours from
					Myawaddy
10.	Mom	09/03/2014	Karen	30s to	Mae La Refugee Camp, originally
				40s	from Karen state.
11	Baitong	18/03/2014	Karen	44	A village near Hpa-An, Karen State

# Other

No.	Name (Alias)	Date	Ethnicity	Age	Occupation
01.	Kanokporn	22/02/2014	Thai – Chinese	32	Work in development organization
O2.	Luxika	08/03/2014	Thai – Chinese	58	Housewife of business man
O3.	Ranee	18/03/2014	Thai – Chinese	58	(Retire) housewife, use to be cilvil servant.
04	Monk	28/03/2014	Thai-Karen	-	Monk

All of them work and live in Bangkok Metropolitan Area

## **Appendix 2: Sample interview questions**

The questions below are the example of interview questions that are used to interview migrants.

### Background information

What is your name?

How old are you?

Which ethnic group do you identify with?

What do your parents do for living?

How long have you worked as domestic workers?

How many people are in your family?

**Migration Process** 

When did you come to Thailand?

Why did you want to come to Thailand?

How did you come to Thailand?

Who assisted you to cross the border? How did you escape the police?

### Working life in Thailand

How did you find your first employers? How did you find your current employers? Please describe your daily routine. What is the characteristic of good employers and bad employers? How many hours you have worked per day? Do you have a free time, holiday or weekend? What do you do when you don't have to work? Who find school for your children? Who do you call when you have problem? What make you feel sad during work? What was the worst thing that happened to you? Have anyone asked you for help? Please tell me about your life! How did you meet your husband? Why did you trust this particular person? Why did you asked this person for help?

#### Future

Do you plan to go back to Burma? When? What is your future plan? Any particular thing you are afraid of?